

WORKING PAPER SERIES NO. 416 / NOVEMBER 2004

EUROSYSTEM INFLATION PERSISTENCE NETWORK

PRICE SETTING BEHAVIOUR IN SPAIN

STYLISED FACTS USING CONSUMER PRICE MICRO DATA

by Luis J. Álvarez and Ignacio Hernando



WORKING PAPER SERIES

NO. 416 / NOVEMBER 2004

EUROSYSTEM INFLATION PERSISTENCE NETWORK

PRICE SETTING BEHAVIOUR IN SPAIN

STYLISED FACTS USING CONSUMER PRICE MICRO DATA '

by Luis J. Álvarez² and Ignacio Hernando³

This paper can be downloaded without charge from http://www.ecb.int or from the Social Science Research Network electronic library at http://ssrn.com/abstract_id=617809.







ublications will carry a motif taken from the €100 banknote.



 This study was conducted in the context of the Eurosystem Inflation Persistence Network. We are extremely grateful to the Instituto Nacional de Estadística for providing us with the dataset, and particularly to Aránzazu García-Almuzara, Ignacio González-Veiga and Alberta Ruiz del Campo for their comments and help. Javier Jareño and Javier Vallés also provided useful suggestions. In addition, we wish to thank an anonymous referee for the ECB working paper series and all the other network members for discussions and suggestions, in particular Luc Aucremanne, Pablo Burriel, Stephen Cecchetti, Emmanuel Dhyne, Jordi Galí, Hervé Le Bihan, Pedro Neves and Patrick Sevestre. Seminar participants at an NBER Summer Institute Workshop and at the Banco de España also provided helpful comments. 2 Banco de España, Alcalá 48, 28014 Madrid, Spain; e-mail: Ijalv@bde.es 3 Banco de España, Alcalá 48, 28014 Madrid, Spain; e-mail: hernando@bde.es

The Eurosystem Inflation Persistence Network

This paper reflects research conducted within the Inflation Persistence Network (IPN), a team of Eurosystem economists undertaking joint research on inflation persistence in the euro area and in its member countries. The research of the IPN combines theoretical and empirical analyses using three data sources: individual consumer and producer prices; surveys on firms' price-setting practices; aggregated sectoral, national and area-wide price indices. Patterns, causes and policy implications of inflation persistence are addressed.

The IPN is chaired by Ignazio Angeloni; Stephen Cecchetti (Brandeis University), Jordi Galí (CREI, Universitat Pompeu Fabra) and Andrew Levin (Board of Governors of the Federal Reserve System) act as external consultants and Michael Ehrmann as Secretary.

The refereeing process is co-ordinated by a team composed of Vítor Gaspar (Chairman), Stephen Cecchetti, Silvia Fabiani, Jordi Galí, Andrew Levin, and Philip Vermeulen. The paper is released in order to make the results of IPN research generally available, in preliminary form, to encourage comments and suggestions prior to final publication. The views expressed in the paper are the author's own and do not necessarily reflect those of the Eurosystem.

© European Central Bank, 2004

Address Kaiserstrasse 29 60311 Frankfurt am Main, Germany

Postfach 16 03 19 60066 Frankfurt am Main, Germany

Telephone +49 69 1344 0

Internet http://www.ecb.int

Fax +49 69 1344 6000

Telex 411 144 ecb d

All rights reserved.

Reproduction for educational and noncommercial purposes is permitted provided that the source is acknowledged.

The views expressed in this paper do not necessarily reflect those of the European Central Bank.

The statement of purpose for the ECB Working Paper Series is available from the ECB website, http://www.ecb.int.

ISSN 1561-0810 (print) ISSN 1725-2806 (online)

CONTENTS

Ał	stract	4
No	n-technical summary	5
1	Introduction	6
2	The dataset	7
	2.1 Specific data issues	8
	2.2 Consumer price developments over 1993-2001	8
3	Statistical issues	10
	3.1 The frequency of price changes and the	
	duration of prices	10
	3.2 Size of price changes	П
	3.3 Aggregation (weighting)	П
4	Frequency of price changes	13
	4.2 Factors that affect the frequency of	
	price changes	13
	4.2.1 Seasonality	14
	4.2.2 Indirect taxation: VAT and excise duties	14
	4.2.3 Macroeconomic conditions:	14
	4.2.3 Macroeconomic conditions: inflation rate	15
	4.2.4 Attractive prices	15
	4.2.4.1 Reasons for the use	
	of attractive prices	16
	4.2.4.2 Relevance of attractive prices	16
	4.2.4.3 Attractive prices and	
	frequency of adjustment	17
	4.2.5 Determinants of the	
	time-series variation of the frequency of price changes	17
5	Size of price changes	19
C	5.1 Main findings	19
	5.2 Factors affecting the size of price changes	19
	5.2.1 Seasonality	19
	5.2.2 Indirect taxation: VAT and	
	excise duties	20
	5.2.3 Macroeconomic conditions:	
		20
	5.2.4 Attractive prices and size	20
	• - <i></i>	20
	5.2.5 Determinants of the time-series variation of the size of	
		21
6	Duration	22
	6.1 Main results	22
	6.2 Probability of changing prices	
	as a function of duration. Logit models	23
		24
7	Conclusions	26
		27
		28
Eu	opean Central Bank working paper series	58

Abstract

This paper identifies the basic features of the price setting mechanism in the Spanish economy, using a large dataset that contains over 1.1 million price records and covers around 70% of the expenditure on the CPI basket. In particular, the paper identifies differences in the frequency and size of price adjustments across types of products and explores how these general features are affected by certain specific factors: seasonality, the level of inflation, changes in indirect taxation and the practice of using psychological and round prices. We find that prices do not change often but do so by a large amount, although there is a marked heterogeneity across products. Moreover, the high frequency of price reductions suggests that there is no strong downward rigidity. Our evidence also supports the use of time and state-dependent pricing strategies.

Keywords: price setting, consumer prices, frequency of price changes. JEL Codes: E31, D40, C25.

Non-technical summary

This paper characterises the basic features of the price setting mechanism in the Spanish economy, drawing on a large dataset containing around 1.1 million price records, covering the period from January 1993 to December 2001. The dataset includes prices for items corresponding to product categories that cover around 70% of the expenditure on the CPI basket and therefore is well suited for the analysis of the key features of price setting behaviour. In particular, the relatively large time dimension of the dataset allows us to explore the relationship between price setting behaviour and macroeconomic conditions, in particular the level of inflation. The main conclusions of our empirical research are the following:

1. Consumer prices are moderately sticky. The average monthly frequency of price changes is 0.15, so that prices change on average once every 6 or 7 months. Alternatively, the direct computation of the duration of price spells indicates that the average duration is slightly over 1 year. The difference between these two statistics is mainly explained by the asymmetry of the cross-item distribution of the frequency of price changes.

2. We find a great deal of heterogeneity in the frequency of price adjustment. As expected, the flexibility of prices is greatest for unprocessed food prices and the highest degree of price stickiness is observed for services. Heterogeneity is somewhat less marked for the size of price changes.

3. While nominal prices are moderately sticky, we do not find signs of a higher degree of downward rigidity. With the exception of the services component, price decreases are only slightly less frequent than price increases, which is consistent with a positive and moderate trend inflation.

4. Even though prices of most products do not change often, they typically change by a large amount (8.6% on average). Moreover, the size of price decreases tends to be somewhat higher than that of price increases.

5. The observed seasonality in the frequency of price changes suggests that some retailers follow time-dependent strategies. The highest frequency of price changes is found at the beginning of the year. In turn, the size of price changes shows a mild seasonal pattern, mostly reflecting the pattern in the size of price increases.

6. Both the frequency and the size of price adjustment depend on the inflation rate. Upward changes in prices are more frequent when inflation is high and downward changes are less frequent with high inflation rates. This fact is consistent with state-dependent pricing strategies. Similarly, as expected, periods of high (low) inflation are characterised by larger (smaller) price increases. Conversely, price decreases are smaller (larger) in high (low) inflation periods.

7. The application of state-dependent strategies is also supported by the fact that the frequency of price changes is significantly affected by changes in indirect taxation. However, these changes do not generally affect the size of price adjustments, a fact that could be due to their small magnitude with regard to the typical size of a price change.

8. Use of attractive prices (i.e. psychological and round) is associated with more sluggish price adjustment and with larger price changes. In absolute terms, price rises and decreases are larger for retailers using attractive pricing strategies.

9. Once we control for individual effects, we do not find a significant relationship either between the duration of a price spell and the probability of observing a price change or between the duration of a price spell and the size of the price change.

1 Introduction

Individual firms do not continuously adjust their prices to all the relevant shocks in the economy. This fact is not controversial and is a standard assumption in macroeconomic modelling that allows for nominal shocks to influence real variables. However, the precise way to incorporate microfounded price adjustment policies in macroeconomic models is less clear-cut. A large strand of literature has been devoted to analysing the implications of alternative forms of nominal rigidities on the dynamic behaviour of aggregate inflation and output. This theoretical literature has shown that the nature of these nominal rigidities determines the shape of the response of the economy to a broad range of potential disturbances and has clear implications for the conduct of monetary policy. In spite of the relevance of the issue of price setting behaviour, empirical work in this area has been scarce and partial, probably due to the lack of appropriate information at the individual level. Moreover, the scant available research has mainly focused on the US economy.

The aim of this paper is to characterise the basic features of the price setting mechanism in the Spanish economy, drawing on a large dataset containing around 1.1 million price records, covering the period from January 1993 to December 2001. The dataset includes prices for items corresponding to product categories that cover around 70% of the expenditure on the CPI basket and therefore is well suited for the analysis of the key features of price setting behaviour. In particular, the relatively large time dimension of the dataset allows us to explore the dependence of price setting behaviour on macroeconomic conditions, especially the level of inflation.

The scope of the paper is basically descriptive. We provide evidence on the frequency and size of price adjustments, distinguishing between price increases and price reductions, and report direct estimates of the duration of price spells. We document differences in the general features of price setting behaviour across types of products and explore how these general features are affected by some specific factors: seasonal patterns in pricing behaviour, the level of inflation, changes in indirect taxation and the practice of using psychological and round prices.

The remainder of this paper is organised as follows. Section 2 describes the dataset and provides a brief overview of the main inflation developments over the sample period. Section 3 discusses the statistical concepts used throughout the analysis. Section 4 presents the main findings on the frequency of price changes, while Section 5 reports the basic results on the size of price changes. Section 6 reports direct estimates of price durations and analyses the influence of the duration of price spells on the probability of adjusting the price as well as on the magnitude of the price change. Section 7 summarises our conclusions.

2 The dataset

The available dataset is a longitudinal subsample of the micro data collected by the Instituto Nacional de Estadística (INE) in order to compute the 1992-based Spanish Consumer Price Index (CPI). The data-collection methodology is described in INE (1994) and the restrictions applied by INE to define the subsample reflect the legal requirement to preserve the confidentiality of the data.

The sample is made up of 1,112,076 prices and contains monthly records for 9 years (January 1993 to December 2001). Each individual price record corresponds to a precisely defined item sold in a particular outlet in a given city at a given point in time. Therefore, individual items can be followed over time within the same outlet. Along with each individual price quote (i.e. the exact¹ price level of the product), the following additional information is provided:

- the year and month of the record
- an item code, which allows the subclass² to which the item belongs to be identified
- a numeric outlet code, which does not enable the name of the outlet to be identified
- a numeric city code, which does not permit the name of the city to be identified
- a numeric code to specify whether the item is a good or a service

In the process of data collection for the computation of the CPI, item substitutions may occur. In such a case, INE, for the purpose of the computation of the index, introduces an adjustment for differences in the characteristics of the sampled products. Unfortunately, we do not have an indicator to signal that a product replacement has taken place. Therefore, our results on the frequency and the magnitude of price changes depend to some extent on the adjustment procedures by the statistical office.

The 1992-based CPI subclass code along with the goods or services code enables us to obtain two additional breakdowns. On the one hand, the components typically used by the Eurosystem to analyse inflation developments (e.g. in the Narrow Inflation Projection Exercise). On the other hand, the harmonised Classification of Individual Consumption by Purpose (COICOP) classification at the 2 digit level.

On the whole, the dataset covers 70.1% of the expenditure on the CPI basket and exclusions mostly correspond to those product categories whose prices are collected in a centralised way, such as housing rents, energy, telecommunications, car prices, tobacco, financial, insurance and household services, hospital and dental services, and hotels. The database excludes administered prices with the exception of those determined at the regional or local level.

Table 2.1 illustrates the breakdown of the available records by HICP subcomponents, and their coverage rate. The coverage rate is relatively uniform across nonenergy components, the highest one corresponding to unprocessed food (88.7%) and the lowest one to processed food (65.6%), reflecting the absence of tobacco prices. Moreover, since all energy prices were regulated in most of the sample period no information is available for them. In terms of the COICOP-2 classification, disparities in the coverage are considerably

¹ Strictly speaking, the information available refers to final prices, so that all correcting coefficients are considered, including those to correct changes in the unit of measure.

² Roughly speaking, this corresponds to the COICOP 5 digit code.

greater. Almost full coverage is available for groups 3 (Clothing and footwear), 11 (Hotels, cafés and restaurants) and 12 (Miscellaneous goods and services). On the other hand, the lowest coverages correspond to groups 2 (Alcoholic beverages and tobacco), 7 (Transport) and 4 (Housing). For group 8 (Communications) no information is available, since their prices were regulated.

The restrictions in terms of coverage reflect the restrictions in terms of items (190 items out of the 471 are available) and subclasses (71 subclasses are available out of 110 in the CPI). Moreover, with respect to the geographical coverage, it has to be stressed that information refers to the 17 regional capitals³, whereas the 1992-based CPI recorded prices in 130 cities for food items and in 70 cities for the rest of the items. Given these data restrictions, a note of caution must be introduced when extrapolating the results to the whole economy.

2.1 Specific data issues

The periodicity of data collection is product-dependent, although prices are generally collected on a monthly basis. On the one hand, some prices like those corresponding to fresh fruit and vegetables are collected three times every month. In these cases INE has provided us with the price corresponding to the first week of every month and has chosen products which are generally available all year round. On the other hand, for some non-food items, prices are generally collected on a quarterly basis using a rotating scheme, so that the same outlet is interviewed only once a quarter. However, if important changes are observed in a given month all outlets are investigated in the next month. In these cases, we have information for those outlets typically interviewed in the first month of every quarter and INE follows the *carry forward* procedure: the unobserved price of the item is assumed to be the same throughout the quarter⁴.

Note that sales and promotion prices were not included in the 1992-based Spanish CPI, so that a lower frequency of price adjustment and smaller sizes of price changes are expected in comparison to countries which considered them. Furthermore, energy prices, most of which change rather frequently, are excluded. Finally, some administered prices are included and they are typically characterised by low frequencies of adjustment.

2.2 Consumer price developments over 1993-2001

For reference purposes it is useful to describe inflation performance in the sample period⁵. Figure 2.1 depicts the year-on-year rate of change of headline inflation and its main subcomponents. The observed variation over time of the inflation rate is useful to unveil potential differences in price setting behaviour at different levels of inflation. Moreover, the different CPI components show different inflation developments.

The sample period can be divided into two subperiods characterised by decreasing and rising inflation. In the 1995-1998 convergence period, the fulfilment of the inflation targets set by the Bank of Spain was facilitated by the increasing credibility of economic policy as a whole and by the change of expectations induced by the perspective of joining EMU. This change was helped, although with a certain delay, by the progressive convergence of wage growth to consumer prices growth. The year on year CPI rate was reduced from around 5% at the start of 1995 to 1.4% at the end of 1998.

The 1999-2001 period was characterised by a rise in the consumer-price inflation rate. In 1999 and 2000, the acceleration of the CPI was largely due to external factors: higher oil and other commodity prices on international markets and the depreciation of the euro

 $[{]f 3}$ There are two exceptions: the regions (Comunidades Autónomas) of Galicia and Extremadura are represented by the cities of A Coruña and Badajoz.

⁴ There are 23 items out of 190 whose prices are collected at a quarterly frequency.

⁵ A more detailed review of the recent inflation developments in the Spanish economy can be found in Malo de Molina (2003).

against the dollar. In contrast, over the course of 2001, the deterioration of inflation responded to domestic factors. In particular, some food prices were particularly affected as a result of the veterinary crises affecting domestic livestock.



3 Statistical issues

The aim of this paper is to characterise the main features of the price setting behaviour of the Spanish economy. For this purpose, we offer a descriptive analysis mainly based on the frequency and size of price adjustments, although we also provide some direct estimates of the duration of price spells. In this section, we provide a brief explanation of the statistics used throughout our analysis, making explicit the assumptions under which they are derived.

3.1 The frequency of price changes and the duration of prices

The periodicity with which prices are changed by firms represents a key element to characterise price-setting behaviour. To quantify this periodicity two different measures may be used: the frequency of price changes and the duration of price spells. Obviously, these two measures are closely linked and should lead to similar conclusions in terms of the degree of price flexibility and, in particular, when establishing comparisons across products.

We define the frequency of price changes as the proportion of non-zero price change observations divided by the total number of observations. The main advantage of this statistic over the average duration of price spells is that it uses all the relevant statistical information available, and consequently, is less potentially affected by selection biases. By contrast, the direct computation of the average duration of price spells has to be done taking into account only uncensored spells (i.e. those spells that start and end with a price change⁶). In addition to discarding some available information, this procedure of computing average duration may be affected by a selection bias since longer price spells are more likely to be censored and, hence, to be discarded. Additional advantages of using the frequency of price changes are that long time series are not required and the measure is less sensitive to specific events. A disadvantage is that it is difficult to derive the full distribution of price durations and for this reason in the last section of the paper we also report the direct computation of price durations, which can be compared to those predicted by theoretical models.

As discussed in Baudry et al. (2004), assuming stationarity and homogeneity of price change behaviour in the cross-sectional dimension, the inverse of the frequency of price changes converges, in a large sample, to the mean duration. This asymptotic property can be used to obtain an estimate of the average duration of price spells (T^F) from the frequency of price changes (*F*). Thus:

$$T^F = \frac{1}{F}$$

This expression is derived under the implicit assumption of discrete time (i.e. prices are changed at most once per month, precisely at the monthly interval). On the other hand, if we assume that prices can be changed at any moment, with the probability of a price change being constant within a given month, the average duration can be derived from the following expression:

$$T^F = -\frac{1}{\ln(1-F)}$$

Working Paper Series No. 416 November 2004

⁶ Alternative definitions of censoring might be used. For instance, Aucremanne and Dhyne (2004) consider as uncensored spells those that start and end with a price change or with a product replacement. In other words, price spells affected by product replacement are considered as uncensored spells. Since we do not have an indicator to signal that a product replacement has taken place we cannot adopt that definition.

We present the analysis of price durations (T^{p}) based on the frequency approach in Section 4, while in Section 6 we report the results based on the direct computation of the duration of the price spells (T^{D}).

3.2 Size of price changes

In addition to the measures of frequency of price adjustments and duration of price spells, we also compute a measure of the size of price changes (ΔP), which is the logarithmic change. Here, our assumption is that prices only change once within a given month (or within a quarter for those products whose prices are collected on a quarterly basis). This assumption seems to be realistic with the exception of some unprocessed food items for which the duration of prices is often shorter than a month⁷.

3.3 Aggregation (weighting)

In order to compute aggregate measures of the variables of interest (frequency of price changes, duration of prices and size of price changes), it seems desirable to employ some weighting scheme. In the cases of the frequency and the size of price changes, we use just one weighting procedure. Since the subclass level is the lowest level for which we have CPI weights, we first compute for each subclass the frequency (or the size) of price changes using all the observations of products belonging to that subclass. We then compute the aggregate frequency (or size) averaging over subclasses using CPI weights.

The computation of an aggregate indicator of price duration might be based either on the frequency approach or on the duration approach. Under the first approach, the simplest alternative would be to make use, at the most aggregate level, of the theoretical relationship between the frequency of price changes and the average price duration. In other words, the overall duration could be computed as the inverse of the weighted average frequency, which is a weighted harmonic mean of the durations at the subclass level:

$$T^{F} = \frac{1}{F} = \frac{1}{\sum_{j} w_{j} F_{j}} = \frac{1}{\sum_{j} w_{j} (1/T_{j}^{F})}$$

where *j* is an index for subclasses and w_j denote CPI weights.

This indicator is not fully consistent with the assumption of homogeneity of observations that is required for the theoretical relationship between the frequency of price changes and the average price duration to hold. An alternative procedure would be first to compute the average duration at the subclass level and then average over all subclasses using CPI weights. Thus, an estimate for the overall price duration would be given by the weighted average of the inverse frequencies, as in the following expression:

$$T^{F} = \sum_{j} w_{j} \frac{1}{F_{j}} = \sum_{j} w_{j} T_{j}^{F}$$

A potential problem with this approach comes from the fact that some frequencies of price changes may be very close to zero, implying very high durations and strongly influencing aggregate duration. Bearing this in mind, the weighted median of inverse frequencies may be a reasonable alternative indicator. This measure, used among others by Bils and Klenow (2002) is less sensitive to the existence of items with very high price durations although, as pointed out by Baudry et al. (2004), it cannot be interpreted as an estimator for the average price duration⁸.

⁷ As mentioned above, INE collects price quotes for these items three times a month.

⁸ Another possibility to ensure robustness would be to compute a trimmed mean.

The computation of an aggregate indicator of price duration stemming from the direct computation of the duration of the individual price spells admits several alternatives. The simplest one is to weight equally the durations of all the individual price spells. As an alternative, we use a weighting scheme in which the weight assigned to each price duration is the CPI weight of its subclass divided by the number of observed durations in that subclass. Thus, we are taking into account that, for a given time horizon, a larger number of spells is observed for products with short durations. However, with this approach we still overweight, within each subclass, trajectories with very short durations. For this reason, we also consider a third weighting approach in which we first average durations within the same trajectory before deriving the distribution at the subclass level.

As mentioned in the previous section, price quotes for some items are collected on a quarterly basis and, therefore, an assumption needs to be made in order to compare the frequency of price changes of these items with that corresponding to items with monthly price observations. In this respect, we follow the *carry forward* approach of INE: for items with quarterly observations, the observed price in one month is assumed to be the same in the following non-observed two months. In other words, we assume that items with quarterly observations experience at most one price change within a quarter, so that the monthly frequency of price changes is one third of the corresponding quarterly frequency. In our view, this assumption is realistic since these items are typically characterised by long price durations⁹. Nevertheless, this assumption might create a slight downward bias in the estimate of the frequency of price changes.

⁹ An alternative assumption to derive the monthly frequency of price changes from a higher frequency is considered in Bils and Klenow (2002) or Dias et al. (2004). In their approach, the assumption adopted is that the monthly frequency of price changes is constant over time and does not depend on duration.

4.1 Main findings

Table 4.1 reports average frequencies of price changes for our whole basket as well as for the main CPI components. The weighted average frequency of price changes for all items taken together is 0.15. Computing the implied duration as the inverse of the weighted average frequency of price durations under the continuous time assumption, prices change, on average, once every 6.2 months. Our estimate is somewhat lower than the estimates available for other European countries. Aucremanne and Dhyne (2004), Baudry *et al.* (2004) and Dias *et al.* (2004) report estimates of 0.17, 0.19 and 0.22 for Belgium, France and Portugal, respectively. The lower frequency of price changes in Spain is to some extent explained by the absence of energy items in our sample, the fact that sale and promotion prices are not considered in our dataset and the inclusion of some administered prices.

Moreover, there is a marked heterogeneity across products in the frequency of price changes. The flexibility of prices is considerably larger for food than for other items. In particular, focusing on the main CPI components, the highest frequency of price changes is observed for unprocessed food: almost one half of these items change prices in any given month. Two factors help explain the flexibility of these prices to market conditions. On the one hand, changes in the weather determine changes in the supply of unprocessed food. On the other hand, these items are typically highly perishable.

The frequency of price changes is much smaller for the other components, although greater for processed food than for non-energy industrial goods - which in our sample are not affected by sales or promotions. The highest degree of price stickiness is observed for services, whose average frequency is 0.06.

The marked heterogeneity in the degree of price flexibility across products is even more evident when looking at the cross sectional distribution of price changes for items at the most detailed level available (190 items in our case). Figure 4.1 displays the distribution of monthly frequencies of price changes for food and for non-food items (66 and 124 items, respectively)¹⁰.

Columns 2 and 3 of Table 4.1 report monthly frequencies of price increases and decreases, respectively, for all items in our basket and split by main CPI components. A first observation arising from these two columns is that price decreases are rather frequent in our sample: almost 40% of the price changes are price reductions, a fact that could be interpreted against the hypothesis of downward nominal rigidity. Again, there are important differences in this pattern across the main CPI components. Thus, while price reductions in services are quite rare (less than 15% of price changes are decreases), in the case of unprocessed food, price increases and decreases are evenly distributed. Moreover, comparing these figures across the main CPI components a clear pattern emerges: the higher is the average frequency of price adjustment, the higher the proportion of price reductions in price changes. This general rule seems to hold also at a more detailed level (see Table A4.1), and a correlation coefficient of 0.6 is obtained.

4.2 Factors that affect the frequency of price changes

The previous section has documented the existence of important differences in the frequency of price changes across products. Even within each main CPI component category there is a substantial dispersion in the frequency of price adjustment (see Figure 4.1 and Table A4.1).

¹⁰ Table A4.1 reports the average frequencies of price changes at the subclass level.

There are several microeconomic factors that could explain such dispersion, including differences in products' characteristics, characteristics of the markets in which they are sold or shares of the different types of outlet. Regrettably, our database does not allow the potential role of these factors to be explored.

So far, the evidence presented has shown a marked heterogeneity in the degree of price stickiness across products. Moreover, although to a lesser extent, the frequency of price changes also displays variation over time. Figure 4.2 illustrates this fact by plotting the time series of the frequency of price increases and decreases for the whole basket and for the main CPI components. These time series are computed in the cross-sectional dimension as a weighted average of the frequency of price changes at the subclass level¹¹. As the prices of some products in the non-food categories are collected on a quarterly basis we plot the time series at that frequency. In the light of this figure, we explore in this section the relevance of some factors potentially affecting the frequency of price adjustment over time. In particular we focus on four issues. First, we check whether there is a seasonal pattern in the pricing behaviour, that could be considered as evidence of time dependent pricing strategies. Second, we assess the impact of changes in indirect taxation on the frequency of adjustment. Third, to capture possible state-dependent pricing strategies we analyse whether the frequency of price adjustment depends on the inflation level. Finally, we check whether the fact of setting prices in attractive terms (i.e. the use of both psychological and round prices) has a bearing on the frequency of price changes.

4.2.1 SEASONALITY

Figure 4.3 summarizes the seasonal pattern in the frequency of price increases and decreases. It reports a box plot of the frequencies observed in each quarter. The seasonal pattern is clearer in the case of price increases, as was also the case in Figure 4.2. On average, prices are more frequently raised in the first quarter of the year, whereas there are less price changes over the summer months. This evidence can be interpreted as implying that some firms follow time dependent price setting strategies. However, this interpretation has to be qualified by the fact that this seasonal pattern found in retail prices might be the result of a similar seasonal pattern in input prices. In such a case, the seasonal pattern found would be consistent with state-dependent pricing policies (i.e. prices change in reaction to cost shocks).

The seasonal pattern in the frequency of price changes differs across the main CPI components. In general, the seasonal pattern of the frequency of price increases differs from that corresponding to price decreases. In the unprocessed food category the highest frequency of price increases is observed in January, whereas the lowest one is found in February. The opposite result is perceived for price decreases. In the case of processed food products the highest frequency of price increases is observed at the beginning of the year, whereas the highest frequency of price decreases is found over the spring months. In non-energy industrial goods, the lowest frequency of price changes (and, in particular, of price increases) is detected over the summer months, a period in which production of these goods is markedly lower than in the rest of the year. In services, the highest frequency of price changes is found in the first quarter, this being particularly clear in the case of price rises. Again, the lowest frequency is observed in the third quarter.

4.2.2 INDIRECT TAXATION: VAT AND EXCISE DUTIES

During the sample period, certain changes in indirect tax rates were implemented. First, in January 1995 a 1 percentage point increase in VAT rates was introduced. Table 4.2 compares the average frequency of price increases and decreases in January 1995 with the

¹¹ For the analysis of the size of price changes, we follow the same approach.

average of the January figures for the other years in the sample. From this table as well as from Figure 4.2, it can be seen that this tax change had a sizable impact on the frequency of price increases for all categories with the exception of unprocessed food. However, the percentage of firms that changed their prices in that particular period is still relatively low, behaviour that could indicate that the size of the tax change was small in respect to the typical size of price changes.

The sample period was also characterised by certain changes in excise duties affecting tobacco and alcoholic beverages. Unfortunately, our sample does not contain information on tobacco prices, so we restrict our attention to a number of changes in tax rates of excise duties on alcoholic beverages. In particular, tax rates on beer were increased in January 1994, January 1995 and January 1996. In these same months as well as in August 1996 tax rates on spirits were raised. In the case of beer, the frequency of price changes was substantially higher in the months in which the tax changes were implemented, although the frequency of firms changing their prices did not reach 50% (see Figure 4.4). Interestingly, this result is not found in the case of spirits where the tax changes hardly affect the frequency of tax changes. If any, the maximum impact is observed one month after the change in the tax rate.

4.2.3 MACROECONOMIC CONDITIONS: INFLATION RATE

Macroeconomic conditions are likely to influence price setting behaviour. In particular, to shed some light on the importance of state-dependent pricing strategies we could analyse the relationship between the frequency of price adjustment and the level of inflation, measured as the year on year rate of change. This analysis might offer some insights into the determinants of price setting behaviour. Thus, the higher the costs of changing prices, the higher will be the dependence of the frequency of price changes on the average level of inflation.

To obtain a clearer view of the relationship between the frequency of price adjustment and aggregate inflation Figure 4.5 plots, on a quarterly basis, the frequency of price changes, increases and decreases against the average inflation rate. These series are adjusted for seasonality and indirect tax changes¹². This figure indicates that upward changes in prices are more frequent when inflation is higher¹³. Analogously, downward changes in prices are less frequent with high inflation rates. If we jointly consider upward and downward changes, we find that the positive relationship between frequency and inflation dominates.

As can be seen in Figure 4.6, which plots the frequency of price increases (decreases) against the average sectoral inflation rate, similar patterns are observed for the main CPI components. Nevertheless, the negative relationship between the frequency of price reductions and the average inflation rate is less clear in the cases of unprocessed food and services.

In order to obtain additional evidence on state-dependent pricing strategies, we have also considered some other demand pressure variables (measures of the output gap, real GDP and consumption growth) and measures of costs (unit labour costs and compensation per employee), although none of them has been significant in time series regressions.

4.2.4 ATTRACTIVE PRICES

¹² More precisely, Figure 3.6 plots the residuals of the linear regression of the inflation rate on a constant, seasonal dummies and a dummy for the VAT change in January 1995 against the residuals of the linear regression of the frequency of price increases (decreases) on a constant, seasonal dummies and the 1995 VAT change. A similar plot will be made to relate the size of the price changes to inflation.

¹³ We are considering the current year-on-year inflation rate, which is a lagging indicator of month-on-month changes. To try to pick up forward looking elements in the frequency of price change we have run regressions with leads on inflation, although they have not been significant.

4.2.4.1 Reasons for the use of attractive prices

A number of studies have documented the common observation that certain digits are more likely than others to appear as rightmost digit or ending of consumer prices. In particular, the digits 0, 5 and 9 are considerably more frequent in practice than a uniform distribution would suggest. Table 4.3 presents some international evidence in this sense. This tendency is the result of different factors lying behind the behaviour of both consumers and price setters.

Among other factors, the use of so-called round prices (typically those ending in 0 or 5) reflects a more general pattern underlying the use of numbers in human communication (Dehaene and Mehler (1992)), that is associated with the higher cognitive accessibility of these numbers, particularly in decimal numeration systems. Against this background, the use of round numbers facilitates the communication of retailers with consumers (Schindler and Kirby (1997)). Moreover, given the incentives to make transactions easier, the denominations of coins and banknotes used by central banks is an additional factor supporting the use of round prices.

Use of so-called psychological prices is also widespread, reflecting a series of factors. Historically, its use is associated with Macy's department store practice of using 9ending prices to oblige clerks to use the cash register to make change and thus reduce the opportunity to pocket the payment. More recently, other explanations have been put forward. For instance, Schindler and Kirby (1997) propose the perceived gain effect: a tendency for consumers to interpret 9 ending prices as round-number prices with a small amount given back. Moreover, theories of bounded rationality consider that the analysis of every digit in a price is costly, so that rational consumers may decide to ignore rightmost digits. This truncation strategy would result in a benefit to sellers using 9 endings. Finally, Schindler (1991) considers that the price ending serves as a cue that gives information about a product or retail outlet. In particular, just-below price endings may connote sales or discount prices or even that the price is low relative to competition.

4.2.4.2 Relevance of attractive prices

Figure 4.7 offers some evidence on the relevance of attractive price setting strategies of Spanish retailers. Attractive prices defined as those ending in 0, 5 or 9 represent 61% of all prices in our sample. Consumer prices ending in 0 (35%) and 5 (17%) are the most frequent, whereas those ending in 9 (9%) appear to be less important. There also exist considerable differences across the main CPI components, such prices accounting for 87% of prices in the case of unprocessed food but only 51% in the case of non-energy industrial goods. The corresponding figures for processed food and services are 58% and 68%, respectively. These figures are similar, although slightly lower, than the ones obtained by Álvarez and Jareño (2002) for Spanish food prices. For instance, for unprocessed food prices they find a share of 94%, instead of 87%, whereas for processed food prices their figure is 79% instead of 58%. There are two reasons that could explain this difference in results. On the one hand, our database contains final prices, so that changes in weights or size of the items sold affect the price ending from that moment onwards. On the other hand, sales and promotions prices -which probably are set more often in attractive terms- are excluded from our sample.

Figure 4.7 suggests that defining attractive prices in terms of their rightmost digit is probably too general, since the reasons to support the use of attractive prices need not apply to all of them (e.g. 69 or 45 pesetas are probably not considered attractive). As a narrower definition, we consider attractive those prices ending in 00, 25, 50, 75, 90, 95 and 99 pesetas¹⁴. This definition takes into account the importance of 25 and 75 price endings which are associated with the importance of the 25 peseta coin. Using this narrower definition, attractive prices represent around 33% of all prices in our sample with substantial differences

^{14 100} pesetas represented 60 euro cents.

across main components: unprocessed food (49%), processed food (27%), non-energy industrial goods (26%) and services (48%).

4.2.4.3 Attractive prices and frequency of adjustment

In terms of the frequency of price adjustment the existence of attractive pricing could be seen as a rigidity in the price setting process. Given a certain disturbance, retailers may decide not to reset their prices until a new attractive price point is appropriate. This would result in longer price durations than in the case of retailers not using attractive prices and larger sizes of price changes. Employing the above mentioned definition results in the frequency of price adjustment reported in Table 4.4. For all items, the frequency of price adjustment for attractive prices is 0.13, whereas 0.16 is obtained for non attractive prices. In terms of durations, these figures would correspond to 7.4 and 5.8 months, respectively if we employ the continuous time assumption. For the main CPI components the same pattern is observed and attractive pricing leads to more sluggish adjustments. However, there are strong differences across categories. For instance, the frequency of price adjustment for unprocessed food items is only slightly lower for those prices that are attractive in comparison with the rest. At the other end of the spectrum, the frequency of change for non-attractive non-energy industrial goods prices is 60% higher than for the rest. It has to be stressed, though, that these differences may be somewhat underestimated to the extent that the prices in our dataset include the effect of changes in the units of measurement of products sold. Analysis of price increases and decreases shows similar features.

4.2.5 DETERMINANTS OF THE TIME-SERIES VARIATION OF THE FREQUENCY OF PRICE CHANGES

As has been emphasised in the preceding sections, we have found differences both in the cross section and time series variation of prices changes. In this section we concentrate on the time series variation of the frequency of adjustment, given that some of the preceding explanatory factors either depend on time (seasonality), are presumed to affect only specific time periods (changes in indirect taxation) or have an aggregate nature (inflation). This set up also allows us to shed some light on the issue of price versus state dependence of pricing strategies and the impact of attractive pricing behaviour.

To summarise the influence of these factors on the frequency of price changes, we run time-series regressions, both for the whole basket as well as for the CPI components. We estimate the following regression model:

$$F_{t} = \boldsymbol{\alpha} + \sum_{i=1}^{3} \boldsymbol{\beta}_{i} \ \boldsymbol{Q}_{i} + \boldsymbol{\gamma} \ DT95 \boldsymbol{Q}1 + \boldsymbol{\delta} \ INF_{t} + \boldsymbol{\varphi} \ ATR_{t} + \boldsymbol{\varepsilon}_{t}$$

where *F* denotes the average frequency of price changes (alternatively *F*⁺ and *F*⁻ will denote, respectively, the average frequency of price increases and decreases), *INF* is the year-on-year overall CPI inflation rate (or the inflation rate of the corresponding CPI component in the case of the sectoral regressions), *Q* are seasonal dummies¹⁵, *DT95Q1* is a dummy variable for the VAT rate change, *ATR* is the average incidence of attractive prices and ε is a residual term¹⁶. Models are estimated by maximum likelihood.

Table 4.5 reports the results of these regressions. These results may be summarised as follows. First, seasonal dummies are always significant. Seasonal factors significantly affect the average frequency of price changes (both increases and decreases) for the whole CPI basket as well as for the different CPI components. Effects are stronger for price increases

¹⁵ Qi takes value 1 in quarter i and -1 in the fourth quarter. Thus, the constant term can be interpreted as the average frequency and the coefficient of Qi as the difference between the average frequency in quarter i and the overall average frequency.

¹⁶ To account for possible autocorrelation of the residuals we consider Ljung-Box statistics and simple and partial correlograms. Although, we do not restrict a priori the ARIMA model generating the disturbance term we find that, when there is autocorrelation, an AR(1) process is sufficient. Its coefficient is reported as RHO.

than for price decreases and are particularly marked for services. Second, the change in the VAT rate in January 1995 significantly affected the frequency of price increases (with the exception of the unprocessed food component) although it did not have any influence on the frequency of price decreases. Third, inflation has a positive and significant impact on the frequency of price increases (the exception being the processed food component). Nevertheless, the magnitude of this effect is moderate: a one percentage point increase in inflation increases the frequency of price increases that is 0.09). Similarly, inflation has a negative effect on the frequency of price reductions (although this effect is not significant in the cases of unprocessed food and services). This latter result suggests that when inflation is high, firms with a negative gap between optimal and current price have some incentives to delay their price adjustment, expecting that this gap will be closed as a consequence of trend inflation. Finally, a weak negative link between the frequency of attractive prices and the frequency of price changes is found suggesting that those items whose prices are fixed in attractive terms have a lower probability of being changed.

In this section we present results on the size of price changes (ΔP), which are defined as logarithmic changes. Here, our assumption is that prices change only once within a given month (or within a quarter for those products whose prices are collected on a quarterly basis). Note that in the case of some unprocessed food items, in which the duration of prices is often shorter than a month, the computed change is the sum of those actually made within a given month.

5.1 Main findings

Table 5.1 reports the average absolute size of price changes for the whole basket, as well as for the main CPI components. Results for average price increases and decreases are also reported. The size of the typical change in prices is 8.6%, which is relatively large. Moreover, the average size of price increases is 8.2%, and that of price reductions is even larger: -10.3%. Our estimates are similar to the estimates available for other European countries. For positive price changes, Aucremanne and Dhyne (2004), Baudry et al. (2004) and Dias et al. (2004) report estimates of 6.8%, 12.5% and 8.1% for Belgium, France and Portugal¹⁷, respectively, whereas average price decreases are -8.7%, -10,0% and -7.5%. Note that inclusion of sale and promotion prices in our dataset would increase the average absolute size of price changes whereas the consideration of unregulated energy prices would tend to reduce the size of both positive and negative changes. Inclusion of some regulated prices should probably result in larger sizes of price changes.

Again, there is a marked heterogeneity across products in the size of price changes, although it is less pronounced than in the case of the frequency of price adjustment. The highest average size of price changes corresponds to unprocessed food prices, which tend to increase by 14.9% and decrease by 15.6%. Note that changes in weather conditions determine notable changes in the supply of unprocessed food and that their relatively low demand elasticity make supply shifts cause relatively large changes in prices. Moreover, these items are typically highly perishable, so price reductions are more likely the closer to the end of the life of the product. For services, whose prices tend to be sticky, price increases (8.2%) and, particularly, price decreases (-11.2%) are also quite large. Finally, the lowest changes correspond to non energy industrial goods and processed food prices.

5.2 Factors affecting the size of price changes

The previous section has shown the existence of marked differences in the size of price changes across product categories. Moreover, although to a lesser extent, the size of price changes also displays time variation. Figure 5.1 illustrates this fact by plotting the time series of the absolute size of price increases and decreases for the whole basket. As in the case of section 4.2, where we analysed the frequency of price adjustment, we focus on some factors that could affect the size of price changes over time. In particular: the seasonal pattern, the impact of changes in indirect taxation, the level of inflation, and, finally, the incidence of attractive price-setting price strategies.

5.2.1 SEASONALITY

Figure 5.1 shows that the size of price changes displays a mild seasonal pattern, although it is considerably less pronounced than that of the frequency of price adjustment. The seasonal pattern is more marked in the case of price increases, being unclear in the case of price

¹⁷ Estimates for Portugal and Belgium refer to median price changes. Median sizes tend to be smaller than mean sizes.

reductions. On average, price increases tend to be greater in the second half of the year than in the first half. Note that sale prices are not included in our dataset, a fact that would entail a more pronounced seasonal pattern.

Analysing the main CPI components, we find again that the seasonal pattern of the size of price increases is clear, although not in the case of price decreases. As expected, the seasonality is particularly marked in the unprocessed food category and also, but to a lesser extent, in the case of processed food products. The seasonal pattern is much less pronounced in the case of non-energy industrial goods and services, although the size of price increases for services in the fourth quarter is considerably higher than in the rest of the year.

5.2.2 INDIRECT TAXATION: VAT AND EXCISE DUTIES

Figure 5.1 also allows the impact of the 1 percentage point increase in VAT rates that was introduced in January 1995 to be examined. It can be seen that this tax change did not have much impact on the size of price increases or decreases. The relatively small change in the tax rate relative to typical price changes could explain this result. Analysis of the excise duty changes in the sample period shows that in the case of beer the size of the price changes was not affected by the tax rate changes. For spirits, changes in excise duties did not affect the size of price changes, with the exception of the change in January 1996.

5.2.3 MACROECONOMIC CONDITIONS: INFLATION RATE

Macroeconomic conditions are likely to influence price setting behaviour. In particular, to shed some light on the importance of state-dependent pricing strategies we analyse the relationship between the size of price adjustments and the level of inflation. For price increases, we would expect that periods with high inflation would be characterised by larger price increases and vice versa. On the other hand, price decreases should be smaller in high inflation periods and higher in low inflation periods.

To obtain a clearer view of the relationship between the size of price changes and aggregate inflation, Figure 5.2 plots the quarterly time series of the size of price changes, increases and decreases against the average inflation rate. Both series are adjusted for seasonality and indirect tax changes in the way described in section 4.2.3. This figure seems to suggest that the size of price increases is greater the higher the rate of inflation. As expected, for the size of price decreases, we find that periods of higher inflation are associated with smaller –in absolute terms- price decreases.

As can be seen in Figure 5.3, which plots the frequency of price increases (decreases) against the average sectoral inflation rate, similar patterns are observed for the main CPI components. Note that the negative relationship between sectoral inflation and size of price decreases seems to be quite weak.

5.2.4 ATTRACTIVE PRICES AND SIZE OF ADJUSTMENT

As already mentioned, the existence of attractive pricing could be seen as a source of rigidity in the price adjustment process. Thus, retailers using these pricing strategies may decide to delay their response to a given shock until a new attractive price point is appropriate. This would result in price changes being less frequent but of a greater -in absolute termsmagnitude than in the case of retailers not using attractive pricing policies.

As can be seen in Table 5.2, for all items, the average absolute size of a price change is higher for retailers using attractive pricing strategies than for those that do not employ them. Specifically, the typical price change for a firm setting attractive prices is 9.5% whereas it is 8.1% for the rest of retailers. For the main CPI components the same pattern is observed and attractive pricing leads to larger adjustments. The largest discrepancies

correspond to services and unprocessed food prices and the smallest ones to processed food prices. Again, it has to be stressed that these differences may be somewhat underestimated to the extent that the prices in our dataset include the effect of changes in the units of measurement of products sold, a fact that tends to reduce the relevance of attractive prices.

Table 5.2 also reports the size of price increases and decreases for retailers using attractive price setting strategies and the rest. As expected, we observe that attractive pricing strategies lead to greater positive and negative changes. For the main CPI components the same pattern is observed.

5.2.5 DETERMINANTS OF THE TIME-SERIES VARIATION OF THE SIZE OF PRICE ADJUSTMENT

We explore in this section the effect of seasonality, changes in indirect taxation, inflation and attractive price setting strategies on the size of price changes. It has to be stressed that, in contrast to the other determinants, the impact of attractive pricing, should have a greater effect in the cross sectional dimension. Moreover, our price data are somewhat affected by changes in the unit of measure.

To summarise the influence of these factors on the frequency of price changes, we run time-series regressions, both for the whole basket as well as for the CPI components. We estimate the following regression model:

$$S_{t} = \boldsymbol{\alpha} + \sum_{i=1}^{3} \boldsymbol{\beta}_{i} \ \boldsymbol{Q}_{i} + \boldsymbol{\gamma} \ DT95 \boldsymbol{Q}1 + \boldsymbol{\delta} \ INF_{t} + \boldsymbol{\varphi} \ ATR_{t} + \boldsymbol{\varepsilon}_{t}$$

where S denotes the average size of price changes in absolute terms (alternatively S+ and S- denote, respectively, the average size of price increases and decreases), and the rest of the variables have been defined in section 4.2.5. Models are estimated by maximum likelihood¹⁸.

Table 5.3 reports the results of these regressions and results may be summarised as follows. First, for the whole CPI seasonal factors affect significantly the average size of price increases, although not price decreases. By component, there is some heterogeneity and, as expected, the greatest effects are found for unprocessed food prices. There is also a marked seasonal component for price increases in the service sector. Second, the rise in the VAT rate in January 1995 did not significantly increase the size of the price adjustments. Third, inflation has a positive and significant impact on the size of price changes and prices increases, but does not seem to affect the size of price decreases. By component, the same pattern is observed for price increases, except for unprocessed food where sectoral inflation does not have explanatory power. For price decreases and absolute values, inflation does not have much explanatory power. Finally, the size of prices increases is greater the larger is the share of attractive prices and the same pattern is observed –in absolute terms- for price decreases. By component, the evidence is weaker.

¹⁸ We find that an AR(1) or AR(2) process is sufficient to pick up the autocorrelation present in some cases.

6 Duration

As already mentioned in section 3, the main disadvantage of using the frequency approach to analyse the duration of price spells is the difficulty in deriving the distribution of price durations. For this reason, in this section we report direct estimates of price durations. After describing the main features of the distribution of price durations, we analyse the influence of the duration of price spells on the probability of adjusting the price as well as on the magnitude of the price change.

6.1 Main results

Figure 6.1 plots the unconditional distribution of price durations while Table 6.1 summarises the main descriptive statistics computed using alternative weighting procedures. Panel A of Table 6.1 reports the summary statistics of the unweighted distribution. This distribution is extremely skewed and there is a high concentration of price spells with very short durations: the overall mode of the distribution is at duration 1 month and the median is 2 months. The average unweighted duration is 4.9 months and there is a very long right-side tail. To take into account that price spells with very short durations are overrepresented in this distribution (since for a given time horizon, a larger number of spells are observed for products with short durations), we consider two alternative weighting schemes. In the first approach, the weight assigned to each price duration is the CPI weight of its subclass divided by the number of observed durations in that subclass. This weighting procedure results, as reported in Panel B, in higher statistics for durations. The average duration is 11.4 months and the median is 8 months. Nevertheless, with this approach we still overweight, within each subclass, trajectories with very short durations. For this reason, we consider a second weighting approach in which before deriving the distribution at the subclass level, we first average durations within the same trajectory. As reported in Panel C, the average duration is 14.7 months while the median duration is one year.

Finally, it is worth mentioning that the results reported in Figure 6.1 and in Table 6.1 are obtained using uncensored spells only. Nevertheless, the distribution of price durations is quite similar to that constructed using all price spells. If anything, considering censored spells as well moves the distribution of durations to the right, since long-lasting price spells are more likely to be censored. Thus, the weighted average duration of price spells is 14.7 months when censored durations are discarded, compared to 15.7 when all spells are considered.

Figure 6.2 plots the distributions of price durations for the main CPI components. As expected, there are substantial differences across them. In the case of food items, there is a very high concentration of price spells with very short durations. In both cases, processed and unprocessed food, the overall mode of the distribution is at duration 1 month and the 25th percentile of the unweighted distribution is also 1 month. Focusing on the CPI weighted distributions of price spells averaged by trajectory (panel C), the average duration for unprocessed food is 3 months and the 75th percentile is even below this figure.

In the case of non-energy industrial goods and services local modes are observed at months 3, 6, 9, 12 ...reflecting the fact that for some items within these components, prices in our dataset are collected on the first month of each quarter. Nevertheless, the existence of a second mode at 12 months suggests that there is a significant portion of firms, especially in the service sector, that review their prices once a year, in line with a Taylor-type price setting rule. The weighted average durations (panel C) for these two components are well above one year.

There is some evidence suggesting that the duration of a price spell is shorter after a price decrease than after a price increase. Higher frequencies for very short durations are observed in the case of the distribution of durations after a price decrease. The unweighted average duration of price spells after a price decrease is 3.6 months, compared to 5.9 for the distribution of durations after a price increase.

6.2 Probability of changing prices as a function of duration. Logit models

As a way of supplementing the evidence presented in section 4.2, on the determinants of the frequency of price changes, we present estimates of logit¹⁹ models for the probability of changing prices. Unlike the regressions in section 4.2.5 where only the time-series variation was exploited, we now use both the time-series and cross-section dimensions of the data. Thus, we can assess to what extent the duration of a price spell has some bearing on the probability of its being changed. The dependent variable is a binary variable indicating whether a price change has taken place. The explanatory variables are dummy variables that indicate the month, the year²⁰, the type of good and whether the price is set in attractive terms, another dummy variable that takes a value of 1 in January 1995 (when a change in VAT rates took place), the inflation rate (defined as the year-on-year inflation rate of the overall CPI or, alternatively in the logit models for the CPI components, of the corresponding CPI component) and a variable that specifies the number of periods elapsed since the last price change. The model is estimated on the whole sample and observations are unweighted.

The results are reported in the first three columns of Table 6.2. Most of the explanatory variables included in the specification turn out to be significant. Monthly dummies are jointly significant suggesting that seasonal factors have an influence on the probability of changing prices and indicating time dependent strategies. The probability of a price change is almost 40 pp. higher for unprocessed food items (12 pp. higher for processed food and 1.2 pp. higher for non-energy industrial goods) than for services items (services is the reference sector). We find that the higher is the duration of the price spell, the lower is the probability of observing a price change. In particular, an increase of one month in the duration of a price spell decreases the probability of a price change by 0.5 pp. These results are consistent with Baudry et al. (2004) where decreasing hazard functions are reported. The coefficients of the VAT change dummy and the inflation rate are both positive and significant, while that of the attractive pricing dummy is negative and significant. The last six columns of Table 6.2 report the estimates of logit models for the probabilities of a price increase and of a price decrease. The main differences with respect to the results of the logit model for the probability of a price change are the following. First, as expected, the level of inflation has a positive and significant effect on the probability of observing a price increase, but a significantly negative one on the probability of a price decrease. Second, the VAT tax change significantly raises the probability of a price increase, while its effect on the probability of a price cut is substantially smaller.

One potential drawback of the estimates reported in Table 6.2 is that observations belonging to the same trajectory (sequence of prices of a given product in a given outlet) may have special unobserved characteristics that are potentially correlated with the explanatory variables. In our model, the individual effect is likely to be correlated with the duration variable: trajectories with a higher constant term (fixed effect) are likely to display lower durations. In order to control for the existence of such individual effects, we use a fixed effect estimator developed by Chamberlain (1980) and applied in a similar setting by Cecchetti (1986). It is important to note that in the conditional logit estimation procedure proposed by Chamberlain (1980), only the slope parameters are estimated while the constant term for each trajectory is not computed.

¹⁹ Preliminary estimates with probit models showed very similar marginal effects.

²⁰ The results reported correspond to a specification that excludes the year dummies, since we include a macro variable (the inflation rate).

The results of the conditional logit models are reported in Table 6.3²¹. Interestingly. the main difference with respect to the results from the standard logit estimation is found in the coefficients for duration that lose their significance in all cases. The intuitive explanation for this difference is as follows. By construction, the probability of observing a price change is higher in trajectories with shorter durations. If we do not control for individual trajectories, the coefficient of duration will reflect the cross-trajectory dimension in the relationship between duration and probability of price changes. Thus, we find that the longer the duration the lower the probability of finding a price change. However, when we control for individual trajectories, no link between duration and probability of price change is found²². In general, the remaining results do not qualitatively differ from those in Table 6.2. The relevance of time dependent strategies is supported again by the significance of the monthly dummies. The VAT reform induced a substantial increase in the probability of a price change of 13.62, mainly reflecting the rise in the probability of a price increase. The coefficient of the inflation rate is positive and significant, although the magnitude of the effect is moderate: a one percentage point increase in inflation translates into a 0.89 pp. increase in the probability of a price change. Again, when distinguishing between price increases and decreases, the effect of the inflation level is, as expected, positive on the former and negative on the latter. In particular, a one percentage point increase in inflation raises the probability of a price increase by 1.87 pp, approximately three times the effect of inflation on the frequency of upward price changes found in the time series analysis of section 4.2.5. Finally, prices set in attractive terms display a lower probability of being changed.

The results of the logit models with fixed effects for the probability of changing prices for the different CPI components are reported in Table 6.4. In line with the evidence presented in the preceding section, the explanatory power of the different regressors differs across components. Monthly dummies are jointly significant in all cases. In general, duration does not affect the probability of changing prices. A negative effect of duration on this probability is only observed for the processed food component and even in this case the marginal effect is very low. Inflation, measured by the corresponding sectoral index, has a significantly positive effect in the models for all components, the exception being processed food items²³. An additional percentage point of inflation translates into a moderate (between 0.1 and 2.1 pp.) increase in the probability of a price change. The effect of the VAT rate change is significant for all components, with the exception of unprocessed food. The marginal effect of this tax change ranges from 13.6 pp. in the case of non-energy industrial goods to 16.3 pp. in that of processed food. Finally, prices set in attractive terms display a lower probability of being changed.

6.3 Size and duration

In this section, we analyse the determinants of the magnitude of the price changes. In particular, we explore the relationship between the size of the price adjustments and the duration of price spells, which sheds some light on the importance of menu cost models. A positive sign in this relationship would be consistent with the predictions of these models. In the presence of high menu costs, firms will postpone a price change until the desired adjustment is large enough to warrant paying the menu cost. Thus, the higher are the menu costs, the lower will be the frequency of price adjustments and the larger will be the magnitudes of the changes once they take place. Note that a similar argument applies to attractive price setting theories.

²³ A positive effect of inflation is observed in the models for the probability of price increases.



²¹ Two important features of the estimation procedure are worth mentioning. First, trajectories where the dependent variable is either always zero (the price never changes) or always one (the price always changes) are discarded, so the number of usable observations is slightly below that in the standard logit model. Second, variables that do not display within-trajectory variation, such as the dummy variables for the type of good, cannot be included in the model.

²² This result is related to that of Álvarez et al. (2004), where it is analytically shown that the common finding of decreasing unconditional hazard functions may be due to composition effects.

Table 6.5 provides a first piece of evidence on this relationship. It reports the average size of price changes for different duration classes²⁴. The relationship between the duration of price spells and the size of the price changes is not clear. In the case of food items, the absence of a monotonic relationship may be explained by the fact that some of these items, being especially prone to suffer frequent supply shifts, have low demand elasticity and, as a consequence, experience large price changes. At the other extreme, in the case of services longer price spells tend to end with larger price changes, which is consistent with the predictions of menu cost models.

Finally, to assess the influence of duration on the size of price adjustment, we use Heckman's sample selection model. We use this two-equation model to explicitly consider the possibility that some firms may decide to delay their desired price adjustment for different reasons (for instance, to avoid printing new catalogues). Hence, we only observe the magnitude of the desired price change, when a price change takes place. The main equation of the model is the one for the desired price change. We assume that when a price change takes place it coincides with the desired price change²⁵. The second equation is the selection mechanism which models the factors determining that a price change occurs. We consider fixed effects both in the selection mechanism and in the equation for the size of price changes. In particular, in both equations we allow for a different constant term for each specific product in each specific outlet. To be able to estimate the corresponding model we have restricted our sample to those observations corresponding to just one city. Thus, the selection mechanism is a probit model explaining the probability of a price change with the same explanatory variables as in the model of the previous section. We exclude duration, because in the previous section we found it to be insignificant and the dummies for special components and the dummy for attractive pricing²⁶ because they are fixed effects themselves. To specify the model for the size of the price changes, we assume that the size of price changes (in absolute terms) is determined exclusively by the duration of the price spell, the inflation rate and a set of seasonal dummies. The results are reported in Table 6.6. First, we note that $\,
ho$, the correlation coefficient of the disturbance terms of the size and probability of change equations is different from zero, so standard regression techniques applied to the size equation would lead to biased results. Second, it is worth noting that the estimates for the selection mechanism are consistent with the estimates of the logit model in the previous section. Third, as expected, inflation has a significantl positive influence on the size of price adjustments. Interestingly, the estimated coefficient for the duration of the price spell is not significant, indicating that there is not a clear relationship between the time elapsed since the last price change and the magnitude of the price adjustment. This evidence is against the hypothesis of firms facing menu costs.

²⁴ Only uncensored spells are used to compute this table.

²⁵ In other words, the underlying assumption in this approach is that when a firm changes its price, it fully adjusts to the desired price.

²⁶ In principle, outlets could display some variation in their use of attractive prices through time. However, we think that in our sample this observed variation mostly corresponds to the introduction by the INE of correcting coefficients in prices.

7 Conclusions

This paper documents the main stylised facts of the price setting behaviour of Spanish retailers over the period 1993-2001. To this end, we exploit a subsample of the micro prices underlying the 1992-based Spanish consumer price index. Our analysis mainly focuses on the frequency and size of price changes as well as on the duration of price spells. The main conclusions of our empirical research are the following:

1. Consumer prices are moderately sticky. The average monthly frequency of price changes is 0.15, so that prices change on average every 6 or 7 months. Alternatively, the direct computation of the duration of price spells indicates that the average duration is slightly over 1 year. The difference between these two statistics is mainly explained by the asymmetry of the cross-item distribution of the frequency of price changes.

2. We find a great deal of heterogeneity in the frequency of price adjustment. As expected, the flexibility of prices is greatest for unprocessed food prices and the highest degree of price stickiness is observed for services. Heterogeneity is somewhat less marked for the size of price changes.

3. While nominal prices are moderately sticky, we do not find signs of a higher degree of downward rigidity. With the exception of the services component, price decreases are only slightly less frequent than price increases, which is consistent with a positive and moderate trend inflation.

4. Even though prices of most products do not change often, they typically change by a large amount (8.6% on average). Moreover, the size of price decreases tends to be somewhat higher than that of price increases.

5. There is a marked seasonality in the frequency of price changes, which suggests that some retailers follow time dependent strategies. In turn, the size of price changes shows a mild seasonal pattern, mostly reflecting the pattern in the size of price increases.

6. Both the frequency and the size of price adjustment depend on the inflation rate. Upward changes in prices are more frequent when inflation is high and downward changes are less frequent with high inflation rates. This fact is consistent with state-dependent pricing strategies. Similarly, as expected, periods of high (low) inflation are characterised by larger (smaller) price increases. Conversely, price decreases are smaller (larger) in high (low) inflation periods.

7. The application of state-dependent strategies is also supported by the fact that the frequency of price changes is significantly affected by changes in indirect taxation. However, these changes do not generally affect the size of price adjustments, a fact that could be due to their small magnitude with regard to the typical size of a price change.

8. Use of attractive prices (i.e. psychological and round) is associated with more sluggish price adjustments and with larger price changes. In absolute terms, price rises and decreases are larger for retailers using attractive pricing strategies.

9. Once we control for individual effects, we do not find a significant relationship either between the duration of a price spell and the probability of observing a price change or between the duration of a price spell and the size of the price change. This evidence is not consistent with firms facing menu costs.

In future work, we plan to study the price setting behaviour at the producer level and compare it with the price setting behaviour of retailers analysed in this paper. Future research should also focus on testing alternative theories of price setting. To this end, we intend to estimate models that incorporate both state-dependent and time-dependent variables.

References

ÁLVAREZ, L.J., BURRIEL, P. AND HERNANDO, I. (2004): "Do decreasing empirical hazard functions make any sense?", Banco de España, mimeo.

ÁLVAREZ, L. J. AND JAREÑO, J. (2002): "Implicaciones para la inflación de la conversión de precios a euros", Banco de España. Servicio de Estudios. Working Paper No. 0202.

AUCREMANNE, L. AND CORNILLE, D. (2001): "Attractive prices and euro-rounding effects on inflation", National Bank of Belgium Working Paper No. 17.

AUCREMANNE, L. AND DHYNE, E. (2004): "How frequently do prices change? Empirical evidence based on the micro data underlying the Belgian CPI", ECB Working Paper No. 331.

BAUDRY, L., LE BIHAN, H., SEVESTRE, P. AND TARRIEU, S. (2004): "Price rigidity in France. Some evidence from consumer price micro-data", Banque de France, mimeo.

BERGEN, M., CHEN, H. A. AND LEVY, D. (2003): "Making Sense of Ignoring Cents: Another Implication of Rational Inattention", mimeo, Paper presented at the October 10-11 2003 Central European University Workshop, Bucarest.

BILS, M. AND KLENOW, P. (2002): "Some evidence on the importance of sticky prices", NBER Working Paper No. 9069.

CECCHETTI, S. (1986): "The frequency of price adjustment: a study of the newsstand prices of magazines", Journal of Econometrics, 31, 255-274.

CHAMBERLAIN, G. (1980): "Analysis of covariance with qualitative data", Review of Economic Studies, 47, 225-238.

DEHAENE, S. AND MEHLER, J. (1992): "Cross-linguistic regularities in the frequency of number words", Cognition, 43, 1-29

DIAS, M., DIAS, D. AND NEVES, P. (2004): "Stylised features of price setting behaviour in Portugal: 1992-2001", ECB Working Paper No. 332.

FENGLER, M. AND WINTER, J. (2001): "Psychological pricing points and price adjustment in German retail markets", Mannheim Universität, mimeo.

FOLKERTSMA, C. K. (2002): "The euro and psychological prices: Simulations of the worst-case scenario", De Economist, 150(1), 19-40.

HOLDERSHAW, J., GENDALL, P. AND GARLAND, R. (1997): "The widespread use of odd pricing in the service sector", Marketing Bulletin, 7, 53-58.

INSTITUTO NACIONAL DE ESTADÍSTICA (1994): Indice de precios de consumo. Base 1992. Metodología.

MALO DE MOLINA, J. L. (2003): "Una visión macroeconómica de los veinticinco años de vigencia de la Constitución Española", Documento Ocasional No. 0307. Banco de España.

SCHINDLER, R. M. (1991): "Symbolic meanings of a price ending", Advances in Consumer Research, 18, 794-801.

SCHINDLER, R. M. AND KIRBY, P.N. (1997): "Patterns of rightmost digits used in advertised prices:

implications for nine-ending effects", Journal of Consumer Research, 24 (2), 192-201.

Coverage of the sample in terms of expenditure

Main CPI components

	CPI weight	Sample weight	Coverage
Unprocessed food	118.8	105.4	88.7
Processed food	174.8	114.7	65.6
Non-energy industrial goods	328.8	257.8	78.4
Energy	71.2	0.0	0.0
Services	306.4	223.5	72.9
Total	1000.0	701.3	70.1
COICOP groups			
1. Food and non-alcoholic beverages	267.8	215.0	80.3
2. Alcoholic beverages and tobacco	25.8	5.0	19.5
3. Clothing and footwear	114.8	111.1	96.8
4. Housing	102.8	41.1	39.9
5. Furniture and household equipment	64.3	52.7	81.9
6. Health	24.7	20.6	83.2
7. Transport	135.8	37.7	27.7
8. Communications	14.4	0.0	0.0
9. Recreation and culture	67.9	47.1	69.4
10. Education	12.9	8.9	69.3
11. Hotels, cafes and restaurants	109.6	104.3	95.2
12. Miscellaneous goods and services	59.1	57.7	97.6
Total	1000.0	701.2	70.1

Monthly frequency of price changes

requency of price changes	Frequency of price increases	Frequency of price decreases	% of price increases
0.50	0.26	0.23	53.1%
0.18	0.11	0.07	59.0%
0.07	0.05	0.02	74.9%
0.06	0.05	0.01	86.5%
0.15	0.09	0.06	62.2%
	0.50 0.18 0.07 0.06	changesincreases0.500.260.180.110.070.050.060.05	0.500.260.230.180.110.070.070.050.020.060.050.01

Effect of VAT change in the frequency of price changes

Main component	Frequency of	price increases	Frequency of	orice decreases
	January 1995	Other Januarys	January 1995	Other Januarys
Unprocessed food	0,33	0,34	0,16	0,15
Processed food	0,22	0,13	0,07	0,06
Non-energy industrial goods	0,14	0,07	0,02	0,02
Services	0,31	0,13	0,01	0,01
All items	0,24	0,14	0,04	0,04



Frequency of attractive prices (%)

Country	Authors			Price ending	
		0, 5 and 9	0	5	9
New Zealand	Holdershaw et al. (1997)	97	8	29	61
Germany	Fengler and Winter (2001)*	80	-	-	80
United States	Schindler and Kirby (1997)	76	27	19	31
Belgium	Aucremanne and Cornille (2001)	72	-	-	-
Netherlands	Folkerstama (2002)**	67	24	12	31
Spain	Álvarez and Jareño (2002)	91	5	52	34
United States	Bergen et al. (2003)	81	5	11	65
* Result for price	e endings 8 and 9				

** The 5 price ending only covers prices ending in 25 and 75

Frequency of price changes

Main component	Attractive prices	Non attractive prices
Unprocessed food	0.47	0.51
Processed food	0.14	0.20
Non-energy industrial goods	0.05	0.08
Services	0.05	0.06
All items	0.13	0.16

Frequency of price increases

Main component	Attractive prices	Non attractive prices
Unprocessed food	0.25	0.27
Processed food	0.08	0.12
Non-energy industrial goods	0.04	0.06
Services	0.04	0.06
All items	0.08	0.10

Frequency of price decreases

Main component	Attractive prices	Non attractive prices
Unprocessed food	0.22	0.24
Processed food	0.06	0.08
Non-energy industrial goods	0.01	0.02
Services	0.01	0.01
All items	0.05	0.06

										Table 4.5
	Deteri	minants o	of the time	e-series va	ariation o	f the freq	uency of price	changes		
	All ite	ems	Unproces	sed food	Processe	ed food	Non-energy indu	strial goods	Services	
	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value
Intercept	0.141	0.00	0.631	0.00	0.250	0.00	0.060	0.00	0.059	0.00
Q1 Q2	0.010 -0.004	0.00 0.02	-0.018 -0.014	0.00 0.01	0.014 -0.001	0.00 0.67	-0.001 0.005	0.30 0.00	0.035 -0.012	0.00 0.00
Q3	-0.010	0.00	0.034	0.00	-0.015	0.00	-0.022	0.00	-0.016	0.00
DUM95Q1 INF	0.047 0.003	0.00 0.01	-0.005 0.002	0.82 0.03	0.073 0.000	0.00 0.95	0.039 0.003	0.00 0.00	0.071 0.006	0.00
FREQ_ATR	-0.016	0.52	-0.280	0.23	-0.279	0.01	0.001	0.89	-0.061	0.02
RHO					0.495	0.00				
Number of observations	3	5	3	5	35	5	35		3	5
Wald joint significance test p-value	144.02 0		57.		76.06 0		722.0 0	5	668 0	
Log likelihood	127.06		92.		98.4		146.36		128	
Residual standard error			0.01		0.01		0.003		0.00	
Wald sesonality test p-value	48.: 0		47.		20. 0.00		577.52 0		333 0	

Determinants of the time-series variation of the frequency of price increases

	All it	ems	Unproces	sed food	Process	ed food	Non-energy indu	strial goods	Servi	ces
	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value
Intercept	0.084	0.00	0.412	0.00	0.162	0.00	0.041	0.00	0.051	0.00
Q1	0.013	0.00	-0.008	0.25	0.017	0.00	0.002	0.12	0.034	0.00
Q2	-0.008	0.00	-0.023	0.00	-0.011	0.01	0.003	0.00	-0.012	0.00
Q3	-0.009	0.00	0.030	0.00	-0.012	0.00	-0.020	0.00	-0.013	0.00
DUM95Q1	0.044	0.00	-0.005	0.85	0.060	0.00	0.040	0.00	0.070	0.00
INF	0.006	0.00	0.003	0.03	0.004	0.18	0.005	0.00	0.006	0.00
FREQ_ATR	-0.034	0.18	-0.307	0.28	-0.263	0.06	-0.014	0.14	-0.060	0.01
RHO					0.546	0.00				
Number of observations	3	5	3	5	3	5	35		3	5
Wald joint significance test		.13	36.		70.		837.8	2	758	
p-value	(((0		0	
Log likelihood	125.92		85.	11	93.	30	151.8	2	132	.50
Residual standard error	0.00	663	0.02	127	0.01	675	0.003	16	0.00	549
Wald sesonality test	57.		28.		21.		585.7		367	
p-value	((0.00		0		0	

Determinants of the time-series variation of the frequency of price decreases										
	All ite	ems	Unproces	sed food	Process	ed food	Non-energy indu	strial goods	Servi	ices
	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value
Intercept	0.058	0.00	0.220	0.05	0.087	0.00	0.019	0.00	0.007	0.00
Q1	-0.003	0.00	-0.011	0.05	-0.003	0.19	-0.003	0.00	0.002	0.00
Q2	0.004	0.00	0.008	0.10	0.009	0.00	0.002	0.00	0.000	0.82
Q3	-0.001	0.08	0.004	0.43	-0.003	0.17	-0.002	0.00	-0.003	0.00
DUM95Q1	0.003	0.36	0.000	1.00	0.013	0.10	-0.001	0.44	0.001	0.72
INF	-0.002	0.00	0.000	0.61	-0.004	0.00	-0.002	0.00	0.000	0.62
FREQ ATR	0.018	0.10	0.027	0.91	-0.008	0.87	0.014	0.04	-0.001	0.90
RHO	0.010	0.10	0.021	0.01	0.297	0.07	0.347	0.04	0.001	0.00
Number of observations	3	5	3	5	3	5	35		35	5
Wald joint significance test	41.		6.2		44.		172.5	1	38.	
p-value			0.40		0		0		0	
Log likelihood	155	63	93.		120	47	174.7	7	176	97
Residual standard error	0.00		0.01		0.00		0.0016		0.00	
Wald sesonality test	22.		5.2		21.		148.9		35.	
p-value	22.		0.15		0.00		0	-	000.	

Tabl	e	5.	1
------	---	----	---

Size of price changes

CPI component	Average price change	Average price increase	Average price decrease
Unprocessed food	15.2%	14.9%	-15.6%
Processed food	7.3%	6.9%	-8.0%
Non-energy industrial goods	6.5%	6.1%	-8.3%
Services	8.4%	8.2%	-11.2%
All items	8.6%	8.2%	-10.3%



		Table 5.2			
Absolute size of price changes					
Main component	Attractive prices	Non attractive prices			
Unprocessed food	16.2%	14.3%			
Processed food	7.7%	7.2%			
Non-energy industrial goods	7.2%	6.4%			
Services	9.9%	7.4%			
All items	9.5%	8.1%			
Size of price increases					
Main component	Attractive prices	Non attractive prices			
-	-				

Unprocessed food	15.8%	14.0%	
Processed food	7.2%	6.8%	
Non-energy industrial goods	6.8%	5.9%	
Services	9.7%	7.3%	
All items	9.1%	7.7%	

Size of price decreases										
Main component	Attractive prices	Non attractive prices								
Unprocessed food	-16.6%	-14.6%								
Processed food	-8.6%	-7.9%								
Non-energy industrial goods	-8.6%	-8.2%								
Services	-12.7%	-9.3%								
All items	-11.1%	-9.5%								
										Table 5.3
---	--------------------	------------------	--------------------	---------------	--------------------	--------------	-------------------	----------------	--------------------	--------------
				Size of	price cha	nges				
	All it	ems	Unproces	sed food	Process	ed food	Non-energy inc	dustrial goods	Serv	vices
	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value
Intercept	0.0689	0.00	0.1263	0.06	0.1263	0.00	0.0590	0.00	0.0703	0.00
Q1 Q2	-0.0011 -0.0017	0.32 0.08	-0.0004 -0.0024	0.80 0.12	-0.0004 -0.0024	0.45 0.03	0.0008 -0.0021	0.59 0.16	-0.0062 -0.0003	0.01 0.91
Q3 DUM95Q1	0.0009 -0.0065	0.39 0.02	0.0084 -0.0049	0.00 0.47	0.0084 -0.0049	0.58 0.00	0.0015 -0.0064	0.30 0.23	-0.0051 -0.0018	0.02 0.83
INF FREQ ATR	0.0016 0.0393	0.00 0.00	-0.0006 0.0501	0.43 0.71	-0.0006 0.0501	0.61 0.00	0.0016 0.0155	0.19 0.28	0.0031 0.0118	0.07 0.73
RHO1 RHO2	0.2011 -0.4080	0.28 0.02	0.489321	0.04	0.4893	0.00				
Number of observations		6 .46	30		3 74	6	3 14.		-	6
Wald joint significance test p-value	0.	00	44. 0.0	00	0.	00	0.0	02	0.	.32 00
Log likelihood Residual standard error	0.0	160.17 0.0025		.67)65	150.52 0.00		136.23 0.0049		0.0).89 077
Wald sesonality test p-value		4.76 0.19		35.69 0.00		24 16	2.59 0.46			.88 00

			Size of p	orice incre	eases				
All ite	ms	Unproces	sed food	Process	ed food	Non-energy inc	dustrial goods	Serv	/ices
Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value	Coeff.	p-value
0.0669	0.00	0.1380	0.03	0.0477	0.00	0.0535	0.00	0.0714	0.00
-0.0019	0.07	-0.0017	0.39	-0.0010	0.46	0.0009	0.51	-0.0060	0.03
-0.0019	0.04	-0.0064	0.00	0.0009	0.46	-0.0012	0.34	-0.0001	0.97
0.0014	0.14	0.0091	0.00	0.0010	0.41	0.0004	0.76	-0.0048	0.05
-0.0051	0.15	0.0010	0.90	-0.0083	0.07	-0.0053	0.25	-0.0019	0.84
0.0020	0.00	-0.0004	0.53	0.0010	0.01	0.0041	0.00	0.0048	0.01
0.0283	0.04	0.0129	0.92	0.0707	0.00	-0.0114	0.36	-0.0115	0.76
		0.4365	0.05						
36		3	6	3	6	3	6	3	86
48.4	9	41.	.39	35	17	25.	94	35	.14
0.0	0	0.0	01	0.	00	0.0	00	0.	00
151.	11	123	.62	141	.82	141	.55	117	7.22
0.00	32	0.0	071	0.0	042	0.00	042	0.0	085
11.6	9	33	.98	1.	76	1.0	07	21	.11
									00
	Coeff. 0.0669 -0.0019 -0.0019 0.0014 -0.0051 0.0020 0.0283 366 48.4 0.0 151. 0.000 11.6	0.0669 0.00 -0.0019 0.07 -0.0019 0.04 0.0014 0.14 -0.0051 0.15 0.0020 0.00	Coeff. p-value Coeff. 0.0669 0.00 0.1380 -0.0019 0.07 -0.0017 -0.0019 0.04 -0.0064 0.0051 0.15 0.0010 0.00283 0.04 0.0129 0.0283 0.04 0.0129 0.4365 36 3 48.49 41. 0.00 0.0 151.11 123 0.0032 0.00 11.69 33	All items Unprocessed food Coeff. p-value Coeff. p-value 0.0669 0.00 0.1380 0.03 -0.0019 0.07 -0.0017 0.39 -0.0019 0.04 -0.0064 0.00 0.0014 0.14 0.0091 0.00 -0.0051 0.15 0.0010 0.90 0.0223 0.04 -0.0054 0.53 0.0283 0.04 0.0129 0.92 0.4365 0.05 36 36 48.49 41.39 0.01 151.11 151.11 123.62 0.0071 11.69	All items Unprocessed food Process Coeff. p-value Coeff. p-value Coeff. 0.0669 0.00 0.1380 0.03 0.0477 -0.0019 0.07 -0.0017 0.39 -0.0010 -0.0019 0.04 -0.0064 0.00 0.0000 -0.0051 0.15 0.0010 0.90 -0.0083 0.0283 0.04 0.0129 0.92 0.0707 0.4365 0.05 0.011 0.01 36 36 35 35 0.00 0.01 0.01 0.01 151.11 123.62 144 0.0032 0.0071 0.00	Coeff. p-value Coeff. p-value Coeff. p-value 0.0669 0.00 0.1380 0.03 0.0477 0.00 -0.0019 0.07 -0.0017 0.39 -0.0010 0.46 -0.0019 0.04 -0.0064 0.00 0.0009 0.466 0.0014 0.14 0.0091 0.00 0.0010 0.41 -0.0051 0.15 0.0010 0.90 -0.0083 0.07 0.0020 0.00 0.0129 0.92 0.0707 0.00 0.0283 0.04 0.0129 0.92 0.0707 0.00 0.4365 0.05 0.01 0.00 0.01 0.00 151.11 123.62 141.82 0.0032 0.0071 0.0042 11.69 33.98 1.76 0.042 0.042	All items Unprocessed food Processed food Non-energy inc Coeff. p-value Coeff. p-value Coeff. p-value Coeff. 0.0669 0.00 0.1380 0.03 0.0477 0.00 0.0535 -0.0019 0.07 -0.0017 0.39 -0.0010 0.46 0.0009 -0.0019 0.04 -0.0064 0.00 0.0009 0.46 -0.0012 0.0014 0.14 0.0091 0.00 0.0010 0.41 0.0004 -0.0051 0.15 0.0010 0.90 -0.0053 0.07 -0.0053 0.0202 0.00 0.010 0.01 0.0044 0.0004 0.0041 0.0283 0.04 0.0129 0.92 0.0707 0.00 -0.0114 0.4365 0.05 0.01 0.00 0.01 36 36 36 3 35.17 25. 0.00 0.01 0.00 0.01 0.00 0.01	All items Unprocessed food Processed food Non-energy industrial goods Coeff. p-value Cou0 Cou01 Cou01 <td>All items Unprocessed food Processed food Non-energy industrial goods Sen Coeff. p-value p-value Qoodd Qoodd Qooodd Qoodd Qoodd</td>	All items Unprocessed food Processed food Non-energy industrial goods Sen Coeff. p-value p-value Qoodd Qoodd Qooodd Qoodd Qoodd

				Size of p	orice decr	eases				
	All it	ems	Unproces	sed food	Process	ed food	Non-energy in	dustrial goods	Serv	rices
	Coeff.	p-value								
Intercept Q1 Q2 Q3 DUM95Q1 INF FREQ_ATR	-0.0721 -0.0025 0.0004 -0.0018 0.0023 0.0001 -0.0811	0.00 0.31 0.84 0.42 0.78 0.96 0.01	-0.1639 0.0051 -0.0001 -0.0098 0.0045 -0.0002 0.0332	0.00 0.06 0.97 0.00 0.66 0.70 0.77	-0.0474 -0.0012 -0.0009 0.0020 0.0043 0.0001 -0.1260	0.00 0.28 0.36 0.05 0.29 0.91 0.00	-0.0778 -0.0031 0.0080 -0.0051 0.0030 0.0031 -0.0567	0.00 0.46 0.04 0.18 0.83 0.35 0.14	-0.0161 -0.0054 -0.0058 0.0023 0.0297 0.0062 -0.2416	0.70 0.65 0.61 0.84 0.42 0.33 0.05
RHO Number of observations Wald joint significance test p-value Log likelihood Residual standard error Wald sesonality test p-value	16. 0.0 121 0.00 3.5	.09	19 0. 117 0.0 17	.00 .76 00 7.83 084 .13 00	0.6014 36. 38. 0.0 139 0.0 4. 0.0	37 00 0.80 044 33	36. 7. 0.1 101 0.0 5. 0.	10 31 .77 132 04	0.3 68 0.0	34 30 .56 341 14



										Table 6.1
			Dur	ation of p	rice spells					
		Pane	A. Unweight	ed average	e of individu	al price spells	•			
								-		
					Standard	Interquartile	Percentile	Percentile	Minimun	Maximun
	0	bservations	Mean	Median	deviation	range	p25	p75		
All items		167179	4.9	2.0	8.0	4.0	1.0	5.0	1	104
Special components										
Unprocessed food	1	82211	1.9	1.0	2.9	9 1.0	1.0	2.0	1	87
Processed food	2	46920	4.9	2.0	7.0	5.0	1.0	6.0	1	100
Non-energy industrial goods	3	29856	11.3	8.0	11.5	5 11.0	3.0	14.0	1	103
Services	5	8192	13.1	12.0	12.2	8.0	6.0	14.0	1	104

-

Panel B. CPI weighted average of individual price spells

	o	bservations	Mean	Median	Standard deviation	Interquartile range	Percentile p25	Percentile p75	Minimun	Maximun
All items		167179	11.4	8	12.8	11	2	13	1	104
Special components										
Unprocessed food	1	82211	2.0	1	3.4	1	1	2	1	87
Processed food	2	46920	5.1	2	7.4	5	1	6	1	100
Non-energy industrial goods	3	29856	13.7	11	12.3	11	6	17	1	103
Services	5	8192	16.1	12	14.5	14	7	21	1	104

Panel C. CPI weighted average of individual price spells (averaged by trajectory)

	о	oservations	Mean	Median	Standard deviation	Interquartile range	Percentile p25	Percentile p75	Minimun	Maximun
All items		167179	14.7	12.0	13.1	13.4	5.8	19.2	1	104
Special components										
Unprocessed food Processed food Non-energy industrial goods Services	1 2 3 5	82211 46920 29856 8192	3.0 8.1 17.4 20.1	1.6 5.6 14.4 16.0	8.2 11.9	5.8 10.4	1.2 3.5 10.3 11.6	2.6 9.4 20.8 24.0	1 1 1.3	87 100 103 104



L	ogit moo	lel. Cond	litional pr	obability o	f price c	hange. Al	l items			
	P	rice change	s	Pi	rice increase	es	Pr	ice decreas	es	
		Marginal			Marginal		Marginal			
	Coeff.	effect	p-value	Coeff.	effect	p-value	Coeff.	effect	p-value	
Intercept	-2.0995		0.00	-2.2659		0.00	-4.4341		0.00	
month2	-0.0269	-0.0026	0.07	-0.3599	-0.0215	0.00	0.5601	0.0189	0.00	
month3	-0.1300	-0.0122	0.00	-0.3156	-0.0191	0.00	0.2929	0.0089	0.00	
month4	0.0941	0.0095	0.00	-0.1249	-0.0081	0.00	0.4586	0.0148	0.00	
month5	-0.1219	-0.0114	0.00	-0.3923	-0.0231	0.00	0.4214	0.0134	0.00	
month6	-0.2423	-0.0219	0.00	-0.6307	-0.0343	0.00	0.4948	0.0163	0.00	
month7	-0.1862	-0.0171	0.00	-0.4712	-0.0271	0.00	0.4062	0.0129	0.00	
month8	-0.2575	-0.0231	0.00	-0.5168	-0.0292	0.00	0.3288	0.0101	0.00	
month9	-0.1055	-0.0099	0.00	-0.2879	-0.0176	0.00	0.3020	0.0092	0.00	
month10	0.1227	0.0125	0.00	-0.0944	-0.0062	0.00	0.4709	0.0153	0.00	
month11	-0.0082	-0.0008	0.58	-0.1976	-0.0125	0.00	0.3617	0.0112	0.00	
month12	-0.2794	-0.0249	0.00	-0.4631	-0.0267	0.00	0.2081	0.0061	0.00	
spcomp1	2.3719	0.3983	0.00	1.5436	0.1663	0.00	3.3072	0.2997	0.00	
spcomp2	0.9980	0.1186	0.00	0.5366	0.0409	0.00	2.1743	0.1098	0.00	
spcomp3	0.1180	0.0116	0.00	-0.1073	-0.0072	0.00	1.0309	0.0304	0.00	
DUR	-0.0517	-0.0050	0.00	-0.0381	-0.0026	0.00	-0.0701	-0.0019	0.00	
DUM95Q1	0.4967	0.0584	0.00	0.4840	0.0401	0.00	0.0996	0.0028	0.05	
INF	0.0253	0.0025	0.00	0.0650	0.0044	0.00	-0.0447	-0.0012	0.00	
FREQ_ATR	-0.1416	-0.0136	0.00	-0.1524	-0.0101	0.00	-0.0694	-0.0018	0.00	
Number of observations		1082576			1082576			1082576		
Wald joint significance test		180730.99		71387.51			99708.19			
p-value		0			0		0			
Log likelihood		-391061.58			-311190.98			-210993.78		

Table 6.2

Table 6.3

Logit model. Conditional probability of price change. All items. Fixed-effects estimation

	P	rice change	s	Pr	ice increase	es	Pri	ice decrease	es
		Marginal			Marginal			Marginal	
	Coeff.	effect	p-value	Coeff.	effect	p-value	Coeff.	effect	p-value
month2	-0.0272	-0.0068	0.09	-0.3716	-0.0904	0.00	0.5948	0.1416	0.00
month3	-0.1571	-0.0390	0.00	-0.3328	-0.0812	0.00	0.3000	0.0733	0.00
month4	0.1074	0.0268	0.00	-0.1301	-0.0321	0.00	0.4849	0.1167	0.00
month5	-0.1463	-0.0364	0.00	-0.4134	-0.1003	0.00	0.4415	0.1067	0.00
month6	-0.2925	-0.0722	0.00	-0.6654	-0.1575	0.00	0.5155	0.1237	0.00
month7	-0.2285	-0.0566	0.00	-0.5007	-0.1205	0.00	0.4170	0.1010	0.00
month8	-0.3096	-0.0764	0.00	-0.5457	-0.1308	0.00	0.3366	0.0820	0.00
month9	-0.1335	-0.0332	0.00	-0.3076	-0.0752	0.00	0.3062	0.0748	0.00
month10	0.1326	0.0331	0.00	-0.1031	-0.0255	0.00	0.4914	0.1182	0.00
month11	-0.0164	-0.0041	0.30	-0.2106	-0.0518	0.00	0.3765	0.0915	0.00
month12	-0.3330	-0.0820	0.00	-0.4884	-0.1177	0.00	0.2106	0.0518	0.00
DUR	-0.0004	-0.0001	0.37	-0.0004	-0.0001	0.39	-0.0002	-0.0001	0.79
DUM95Q1	0.5529	0.1362	0.00	0.5052	0.1254	0.00	0.1040	0.0257	0.05
INF	0.0358	0.0089	0.00	0.0754	0.0187	0.00	-0.0488	-0.0121	0.00
FREQ_ATR	-0.2727	-0.0678	0.00	-0.2223	-0.0550	0.00	-0.1466	-0.0365	0.00
Number of observations		1068623			1062976			769649	
LR joint significance test		3866.92			4635.71		1384.79		
p-value		0			0		0		
Log likelihood		-322819.64			-271297.86			-172417.83	

Logit r	nodel. Co	nditiona	l probabili	ty of price	change	. Main CPI	l compone	nts. Fixe	d-effects	estimation			
	Unr	processed	food	Pr	ocessed fo	hou	Non-	energy ind	ustrial		Services		
		1000030001	000								00111003		
		Marginal		Marginal			Marginal			Marginal			
	Coeff.	effect	p-value	Coeff.	effect	p-value	Coeff.	effect	p-value	Coeff.	effect	p-value	
month2	0.0601	0.0150	0.05	0.1482	0.0370	0.00	0.0912	0.0223	0.00	-0.80376	-0.11824	0.00	
month3	-0.2139	-0.0533	0.00	0.0162	0.0040	0.55	0.2739	0.0659	0.00	-1.43929	-0.18009	0.00	
month4	-0.0479	-0.0120	0.12	-0.0097	-0.0024	0.73	0.7841	0.1777	0.00	-0.86311	-0.12507	0.00	
month5	-0.0231	-0.0058	0.45	-0.0267	-0.0027	0.33	0.2479	0.0597	0.00	-1.73289	-0.20149	0.00	
month6	0.0197	0.0049	0.52	-0.0997	-0.0248	0.00	-0.1531	-0.0378	0.00	-2.26655	-0.23201	0.00	
month7	0.0077	0.0019	0.80	-0.1352	-0.0335	0.00	0.0010	0.0002	0.98	-1.4389	-0.18005	0.00	
month8	0.2280	0.0569	0.00	-0.1908	-0.0472	0.00	-0.3757	-0.0933	0.00	-2.27439	-0.23239	0.00	
month9	0.2757	0.0687	0.00	-0.1345	-0.0334	0.00	-0.0363	-0.0089	0.28	-1.12721	-0.15271	0.00	
month10	0.1443	0.0361	0.00	-0.0486	-0.0121	0.08	0.7444	0.1696	0.00	-0.81955	-0.12009	0.00	
month11	0.0010	0.0002	0.97	-0.0132	-0.0033	0.63	0.5707	0.1330	0.00	-1.54333		0.00	
month12	-0.2218	-0.0552	0.00	-0.0330	-0.0082	0.23	-0.1303	-0.0322	0.00	-2.23833	-0.23059	0.00	
DUR	0.0024	0.0006	0.09	-0.0016	-0.0004	0.04	-0.0006	-0.0001	0.36	0.001155	0.000206	0.30	
DUM95Q1	0.0829	0.0207	0.21	0.6667	0.1634	0.00	0.5929	0.1362	0.00	0.772445	0.163288	0.00	
INF	0.0050	0.0012	0.00	-0.0027	-0.0007	0.29	0.0858	0.0211	0.00	0.094078	0.016809	0.00	
FREQ_ATR	-0.1493	-0.0373	0.00	-0.2176	-0.0540	0.00	-0.5039	-0.1246	0.00	-0.51178	-0.09078	0.00	
Number of observations		165731			272949			473699			156244		
LR joint significance test		694.3			627.84			4866.89			4577.48		
p-value		0			0		0			0			
Log likelihood		-80356.953	3		-106809.66	6		103409.48	3	-	28793.757		

Table 6.4



		Average size of pr	ico changos hy	duration	Table 6.5
		Average size of pr	ice changes by	duration	
	All items	Unprocessed food	Processed food	Non-energy industrial goods	Services
T<=3	8.8%	15.3%	7.9%	7.4%	7.1%
3 <t<=6< td=""><td>7.7%</td><td>13.8%</td><td>7.5%</td><td>5.8%</td><td>7.0%</td></t<=6<>	7.7%	13.8%	7.5%	5.8%	7.0%
6 <t<=9< td=""><td>7.7%</td><td>13.4%</td><td>7.3%</td><td>6.3%</td><td>6.7%</td></t<=9<>	7.7%	13.4%	7.3%	6.3%	6.7%
9 <t<=12< td=""><td>7.8%</td><td>13.8%</td><td>7.2%</td><td>6.2%</td><td>7.1%</td></t<=12<>	7.8%	13.8%	7.2%	6.2%	7.1%
12 <t<=15< td=""><td>7.8%</td><td>14.6%</td><td>6.9%</td><td>6.4%</td><td>6.8%</td></t<=15<>	7.8%	14.6%	6.9%	6.4%	6.8%
15 <t<=18< td=""><td>8.3%</td><td>13.2%</td><td>7.5%</td><td>6.8%</td><td>8.0%</td></t<=18<>	8.3%	13.2%	7.5%	6.8%	8.0%
18 <t<=21< td=""><td>8.3%</td><td>12.1%</td><td>6.8%</td><td>7.3%</td><td>8.3%</td></t<=21<>	8.3%	12.1%	6.8%	7.3%	8.3%
21 <t<=24< td=""><td>8.6%</td><td>14.5%</td><td>7.1%</td><td>7.1%</td><td>8.5%</td></t<=24<>	8.6%	14.5%	7.1%	7.1%	8.5%
T>24	9.2%	13.7%	7.7%	7.1%	10.1%



Heckman selection model Size equation Coefficient	p-value 0.16 0.00								
Coefficient	0.16								
	0.16								
month2 0.0101									
month2 -0.0121	0.00								
month3 -0.0512									
month4 -0.0214	0.00								
month5 -0.0468	0.00								
month6 -0.0501	0.00								
month7 -0.0426 month8 -0.0634	0.00 0.00								
month9 -0.0266	0.00								
month10 -0.0169	0.00								
month11 -0.0217	0.01								
month12 -0.0611	0.00								
dur -0.0001	0.47								
inf 0.0092	0.00								
Probability of change equation									
Coefficient	p-value								
month2 -0.2072	0.00								
month3 -0.3378	0.00								
month4 -0.0317	0.38								
month5 -0.3361	0.00								
month6 -0.4204 month7 -0.2554	0.00 0.00								
month8 -0.4720	0.00								
month9 -0.2054	0.00								
month10 0.0122	0.73								
month11 -0.1964	0.00								
month12 -0.4418	0.00								
inf 0.0384	0.00								
vat95 0.1784	0.01								
rho 0.8748									
sigma 0.1347									
lambda 0.1178									
Number of observations	59603								
Wald joint significance test	10602.3								
p-value	0								

Table A4.1 Frequency of price changes at the subclass level									
Base 92 CPI subclass	Total	Price increases	Price decreases	%Decreases Total					
Pastry-cooked products	0,17	0,09	0,08	0,44					
Pasta products and other cereal based products	0,13	0,07	0,06	0,47					
Pork	0,40	0,21	0,19	0,47					
Sheep meat	0,54	0,30	0,24	0,44					
Poultry meat	0,41	0,21	0,20	0,48					
Cooked pork meats	0,13	0,08	0,05	0,39					
Canned and processed meat	0,13	0,09	0,05	0,36					
Fresh and frozen fish	0,56	0,29	0,27	0,49					
Dry, smoked and canned and processed fish	0,12	0,08	0,04	0,35					
Fresh and frozen crustaceans and molluscs	0,31	0,17	0,14	0,44					
Liquid milk	0,22	0,13	0,09	0,39					
Cheese and other dairy products	0,16	0,10	0,06	0,39					
Eggs	0,20	0,11	0,09	0,44					
Edible oils	0,37	0,21	0,16	0,44					
Fresh fruit	0,52	0,28	0,24	0,46					
Fresh vegetables	0,61	0,32	0,29	0,48					
Dried vegetables	0,15	0,09	0,06	0,43					
Potatoes and their by-products	0,36	0,20	0,17	0,46					
Sugar	0,16	0,10	0,05	0,35					
Coffee, cocoa, infusions and substitutes	0,18	0,09	0,08	0,47					
Chocolate and chocolate substitutes	0,12	0,08	0,05	0,38					
Other food products	0,15	0,08	0,07	0,44					
Non-alcoholic beverages	0,17	0,10	0,07	0,42					
Spirits Beer	0,21	0,14	0,07	0,34					
Mens outwear	0,15	0,10	0,06 0,01	0,37					
Mens underwear	0,05 0,05	0,04	0,01	0,18					
Womens outwear	0,03	0,05 0,04	0,00	0,09 0,15					
Womens underwear	0,04	0,04	0,01	0,10					
Childrens outwear	0,05	0,03	0,01	0,10					
	0,05		0,01	0,05					
Clothing complements and repairs Mens footwear	0,00	0,05 0,04	0,00	0,05					
Womens footwear	0,05	0,04	0,01	0,10					
Childrens and babies footwear	0,08	0,06	0,01	0,14					
Repair and maintenance of owned housing	0,00	0,05	0,01	0,15					
Water supply	0,08	0,08	0,00	0,03					
Furniture for kitchen and bath	0,05	0,04	0,00	0,17					
Other furniture and decorative ornaments for the	0,05	0,05	0,01	0,11					
Household textiles	0,05	0,04	0,01	0,15					
Refrigerators, washing machines, dishwashers and irons	0,08	0,05	0,04	0,45					
Other Electric Household and Repairs	0,09	0,05	0,04	0,46					
Glassware, tableware, cutlery and their repairs	0,05	0,05	0,01	0,17					
Other kitchen and household equipment and their repairs	0,07	0,05	0,02	0,25					
Goods for cleaning and routine household maintenance	0,14	0,09	0,02	0,40					
Other non-durable household goods	0,11	0,07	0,04	0,37					
Other pharmaceutical products	0,10	0,07	0,04	0,32					
Therapeutic appliances and equipment and repairs	0,08	0,07	0,01	0,16					
Medical, nursing and other out-patient services	0,02	0,02	0,00	0,09					
Health insurance	0,08	0,07	0,01	0,08					
Tyres, Spare Parts and Repairs	0,09	0,08	0,02	0,18					
Other goods in respect of personal transport	0,07	0,06	0,01	0,12					
Local transport	0,07	0,07	0,00	0,02					
Long-distance transport	0,07	0,06	0,00	0,04					
Radio and television equipment and repairs	0,09	0,05	0,04	0,43					
Other audio-visual equipment	0,08	0,03	0,05	0,61					
Photographic equipment, computers and others	0,09	0,03	0,05	0,61					
Games and toys	0,05	0,04	0,01	0,25					
Other recreational goods	0,06	0,03	0,03	0,48					
Cinema, theatre, football and other performances	0,29	0,16	0,13	0,45					
Recreational services	0,04	0,03	0,01	0,32					
Pre-primary and primary education	0,08	0,08	0,00	0,06					
Secondary education	0,08	0,08	0,00	0,02					
Other education	0,06	0,06	0,00	0,03					
Personal Care Services	0,04	0,04	0,00	0,04					
Durable personal care goods	0,12	0,08	0,04	0,34					
Non-durable personal care goods	0,12	0,07	0,04	0,38					
Jewellery, imitation jewellery and their repairs	0,04	0,04	0,01	0,14					

ECB Working Paper Series No. 416 November 2004 43

Size of price changes at the subclass level

Table A4.2

Base 92 CPI subclass	Total (Absolute value)	Total (Mean)	Price increases	Price decrease
Pastry-cooked products	7,9%	0,1%	7,2%	-8,8%
Pasta products and other cereal based products	8,7%	0,1%	8,2%	-9,2%
Pork	9,0%	0,4%	8,9%	-9,1%
Sheep meat	10,8%	0,7%	10,4%	-11,3%
Poultry meat	10,3%	0,5%	10,3%	-10,2%
Cooked pork meats	7,9%	1,3%	7,5%	-8,5%
Canned and processed meat	8,2%	0,9%	7,1%	-10,0%
resh and frozen fish	17,3%	0,1%	17,0%	-17,7%
Dry, smoked and canned and processed fish	7,6%	2,0%	7,4%	-8,0%
resh and frozen crustaceans and molluscs	13,7%	1,1%	13,3%	-14,2%
iquid milk	4,9%	0,6%	4,6%	-5,5%
cheese and other dairy products	8,9%	1,0%	8,1%	-10,2%
ggs	8,7%	0,7%	8,3%	-9,2%
dible oils	6,2%	0,7%	6,2%	-6,3%
resh fruit	16,4%	0,7%	15,8%	-17,1%
resh vegetables	24,9%	0,8%	24,5%	-25,4%
Pried vegetables	9,0%	1,5%	9,2%	-8,7%
otatoes and their by-products	18,1%	1,2%	17,8%	-18,4%
ugar	2,8%	1.0%	2,9%	-2,5%
Coffee, cocoa, infusions and substitutes	7,9%	0,9%	8,4%	-7,4%
Chocolate and chocolate substitutes	7,0%	1,1%	6,6%	-7,7%
Other food products	8,8%	-0,1%	7,8%	-10,1%
Ion-alcoholic beverages	7,4%	0,6%	6,9%	-8,0%
Spirits	4,1%	1,3%	4,1%	-4,2%
Beer	7,7%	1,3%	7,2%	-8,6%
Aens outwear	5,6%	3,1%	5,3%	-7.1%
Aens underwear	6,1%	4,9%	6,1%	-6,8%
Vomens outwear	5,9%	3,7%	5,6%	-7,6%
Vomens underwear	5,9%	4,4%	5,7%	-7,7%
Childrens outwear	6,5%	3,3%	5,9%	-9,7%
Clothing complements and repairs	0,3 <i>%</i> 7,1%	6,4%	5,9 <i>%</i> 7,1%	-9,7 %
Aens footwear	6,5%	3,9%	6,2%	-7,9%
Vomens footwear	6,1%	4,5%	5,9%	-7,7%
Childrens and babies footwear	5,2%		5,9 <i>%</i> 5,2%	-5,4%
	5,2% 6,7%	3,8% 4,5%	5,2% 6,6%	-5,4%
Repair and maintenance of owned housing Vater supply	0,7 % 7,4%		0,0 <i>%</i> 7,2%	-14,7%
Furniture for kitchen and bath		6,5% 4,3%	6,5%	-6,5%
Other furniture and decorative ornaments for the	6,5% 6,4%	4,3% 5,1%	6,4%	-6,3%
Household textiles	0,4% 7,7%	3,8%	6,8%	-0,3%
Refrigerators, washing machines, dishwashers and irons	4,8%	0,1%	4,3%	-5,3%
Other Electric Household and Repairs				
Glassware, tableware, cutlery and their repairs	6,0%	-0,2%	5,5%	-6,7%
	7,7%	4,4%	7,2%	-9,9%
Other kitchen and household equipment and their repairs	8,5%	3,3%	7,9%	-10,6%
Goods for cleaning and routine household maintenance	7,5%	1,2%	7,3%	-7,8%
Other non-durable household goods	8,9%	1,6%	8,3%	-9,9%
Other pharmaceutical products	7,5%	2,8%	7,6%	-7,2%
herapeutic appliances and equipment and repairs	4,3%	2,7%	4,2%	-4,8%
Aedical, nursing and other out-patient services	13,7%	11,1%	13,6%	-15,1%
lealth insurance	6,6%	6,1%	6,9%	-3,2%
yres, Spare Parts and Repairs	5,4%	2,3%	4,7%	-9,0%
Other goods in respect of personal transport	6,5%	3,8%	5,9%	-11,2%
ocal transport	7,1%	6,8%	7,1%	-6,2%
ong-distance transport	3,9%	3,4%	3,8%	-5,0%
Radio and television equipment and repairs	5,7%	0,2%	5,2%	-6,3%
other audio-visual equipment	6,4%	-2,7%	4,7%	-7,5%
hotographic equipment, computers and others	8,0%	-3,7%	5,6%	-9,6%
Sames and toys	7,8%	2,8%	7,1%	-10,1%
Other recreational goods	9,2%	-0,4%	8,5%	-10,0%
Cinema, theatre, football and other performances	8,6%	1,6%	9,4%	-7,7%
Recreational services	7,9%	0,3%	6,0%	-11,8%
Pre-primary and primary education	6,5%	3,7%	5,4%	-25,5%
Secondary education	6,8%	5,6%	6,3%	-34,2%
Other education	5,6%	5,2%	5,6%	-7,5%
Personal Care Services	8,2%	7,5%	8,1%	-9,0%
Durable personal care goods	9,0%	2,5%	8,7%	-9,6%
Non-durable personal care goods	8,9%	1,2%	8,1%	-10,1%
lewellery, imitation jewellery and their repairs	6,7%	5,0%	6,8%	-6,3%

Figure 2.1









Figure 4.2









Figure 4.4





Figure 4.5



Figure 4.6









Figure 5.1











Figure 6.1







European Central Bank working paper series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.int)

- 380 "Optimal monetary policy under discretion with a zero bound on nominal interest rates" by K. Adam and R. M. Billi, August 2004.
- 381 "Fiscal rules and sustainability of public finances in an endogenous growth model" by B. Annicchiarico and N. Giammarioli, August 2004.
- 382 "Longer-term effects of monetary growth on real and nominal variables, major industrial countries, 1880-2001" by A. A. Haug and W. G. Dewald, August 2004.
- 383 "Explicit inflation objectives and macroeconomic outcomes" by A. T. Levin, F. M. Natalucci and J. M. Piger, August 2004.
- 384 "Price rigidity. Evidence from the French CPI micro-data" by L. Baudry, H. Le Bihan,P. Sevestre and S. Tarrieu, August 2004.
- 385 "Euro area sovereign yield dynamics: the role of order imbalance" by A. J. Menkveld, Y. C. Cheung and F. de Jong, August 2004.
- 386 "Intergenerational altruism and neoclassical growth models" by P. Michel, E. Thibault and J.-P. Vidal, August 2004.
- 387 "Horizontal and vertical integration in securities trading and settlement" by J. Tapking and J. Yang, August 2004.
- 388 "Euro area inflation differentials" by I. Angeloni and M. Ehrmann, September 2004.
- 389 "Forecasting with a Bayesian DSGE model: an application to the euro area" by F. Smets and R. Wouters, September 2004.
- 390 "Financial markets' behavior around episodes of large changes in the fiscal stance" by S. Ardagna, September 2004.
- 391 "Comparing shocks and frictions in US and euro area business cycles: a Bayesian DSGE approach" by F. Smets and R. Wouters, September 2004.
- 392 "The role of central bank capital revisited" by U. Bindseil, A. Manzanares and B. Weller, September 2004.
- 393 "The determinants of the overnight interest rate in the euro area" by J. Moschitz, September 2004.
- 394 "Liquidity, money creation and destruction, and the returns to banking" by Ricardo de O. Cavalcanti, A. Erosa and T. Temzelides, September 2004.
- 395 "Fiscal sustainability and public debt in an endogenous growth model" by J. Fernández-Huertas Moraga and J.-P. Vidal, October 2004.
- 396 "The short-term impact of government budgets on prices: evidence from macroeconomic models" by J. Henry, P. Hernández de Cos and S. Momigliano, October 2004.
- 397 "Determinants of euro term structure of credit spreads" by A. Van Landschoot, October 2004.

- 398 "Mergers and acquisitions and bank performance in Europe: the role of strategic similarities" by Y. Altunbas and D. Marqués Ibáñez, October 2004.
- 399 "Sporadic manipulation in money markets with central bank standing facilities" by C. Ewerhart, N. Cassola, S. Ejerskov and N. Valla, October 2004.
- 400 "Cross-country differences in monetary policy transmission" by R.-P. Berben, A. Locarno, J. Morgan and J. Valles, October 2004.
- 401 "Foreign direct investment and international business cycle comovement" by W. J. Jansen and A. C. J. Stokman, October 2004.
- 402 "Forecasting euro area inflation using dynamic factor measures of underlying inflation" by G. Camba-Méndez and G. Kapetanios, November 2004.
- 403 "Financial market integration and loan competition: when is entry deregulation socially beneficial?" by L. Kaas, November 2004.
- 404 "An analysis of systemic risk in alternative securities settlement architectures" by G. Iori, November 2004.
- 405 "A joint econometric model of macroeconomic and term structure dynamics" by P. Hördahl, O. Tristani and D. Vestin, November 2004.
- 406 "Labour market reform and the sustainability of exchange rate pegs" by O. Castrén, T. Takalo and G. Wood, November 2004.
- 407 "Banking consolidation and small business lending" by E. Takáts, November 2004.
- 408 "The great inflation, limited asset markets participation and aggregate demand: FED policy was better than you think" by F. O. Bilbiie, November 2004.
- 409 "Currency mismatch, uncertainty and debt maturity structure" by M. Bussière, M. Fratzscher and W. Koeniger, November 2004.
- 410 "Do options-implied RND functions on G3 currencies move around the times of interventions on the JPY/USD exchange rate? by O. Castrén, November 2004.
- 411 "Fiscal discipline and the cost of public debt service: some estimates for OECD countries" by S. Ardagna, F. Caselli and T. Lane, November 2004.
- 412 "The real effects of money growth in dynamic general equilibrium" by L. Graham and D. J. Snower, November 2004.
- 413 "An empirical analysis of price setting behaviour in the Netherlands in the period 1998-2003 using micro data" by N. Jonker, C. Folkertsma and H. Blijenberg, November 2004.
- 414 "Inflation persistence in the European Union, the euro area, and the United States" by G. Gadzinski and F. Orlandi, November 2004.
- 415 "How persistent is disaggregate inflation? An analysis across EU15 countries and HICP sub-indices" by P. Lünnemann and T. Y. Mathä, November 2004.
- 416 "Price setting behaviour in Spain: stylised facts using consumer price micro data" by L. J. Álvarez and I. Hernando, November 2004.